# BEFORE THE PUBLIC SERVICE COMMISSION OF THE STATE OF MISSOURI



In the matter of the Joint Application	)	
of Gateway Pipeline Company, Inc.	)	Case No. <u>Gm-2001</u> -585
Missouri Gas Company and Missouri	)	
Pipeline Company.	)	

April 2001

# SECOND AMENDMENT TO STOCK PURCHASE AGREEMENT

This Second Amendment is dated as of April 12, 2001, by and among Gateway Pipeline Company, Inc., a Delaware corporation ("Purchaser"), UtiliCorp United Inc., a Delaware corporation ("Seller"), and UtiliCorp Pipeline Systems, Inc., a Delaware corporation (the "Company").

# **RECITALS**

WHEREAS, the parties hereto entered into a Stock Purchase Agreement dated February 1, 2001 (as previously amended on February 16, 2001, the "Original Agreement"); and

WHEREAS, the parties now desire to further amend the Original Agreement.

NOW THEREFORE, in consideration of the mutual promises hereinafter set forth and other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereby agree as follows:

#### **AGREEMENT**

# 1. <u>Amendments</u>.

- a. <u>Section 1.3</u>. The first sentence of Section 1.3 is amended to read in its entirety as follows: "On or before April 12, 2001, Purchaser shall deliver to Seller, by wire transfer of immediately available funds, funds in an amount of \$5,000,000 as an earnest money deposit (the "<u>Deposit</u>")."
- b. <u>Section 1.4</u>. The date "June 30, 2001" in the third line of said Section is hereby amended to read: "September 30, 2001".
- c. <u>Section 2.16</u>. Section 2.16 is hereby amended to add the following new subsection (c):
  - "(c) To the knowledge of Seller, no Environmental Claims are pending or threatened as a result of any environmental conditions or facts disclosed in any of the Environmental Reports set forth on Schedule 2.16(b) hereto."
- d. <u>Section 4.4</u>. Schedule 4.4 to the Original Agreement is hereby amended to read in its entirety as follows:
  - "As exceptions to Section 4.4 of the Agreement, MPC shall have the right to (a) renew the existing agreement No. MP-1011-TAF

with Laclede Gas Company's agent CoEnergy Trading Company, or enter into another agreement with Laclede Gas Company or its agent, with the term of such renewal or agreement not extending beyond December 31, 2002, and provided that any changes to the volume or other terms of such agreement shall be subject to Purchaser's prior written approval, which shall not be unreasonably withheld or delayed, and (b) renew the existing agreement No. MP-1024-TAF with Ameren Services, Co. (d/b/a Union Electric), or enter into another agreement with Ameren Services, Co. (d/b/a Union Electric), with the term of such renewal or agreement not extending beyond December 31, 2002, and provided that any changes to the volume or other terms of such agreement shall be subject to Purchaser's prior written approval, which shall not be unreasonably withheld or delayed."

- e. <u>Section 8.1(b)</u>. The date "December 31, 2001" in the second line of said Section is hereby amended to read: "June 30, 2002".
- f. Section 8.1(c). The phrase "on the 12th Business Day after the execution and delivery hereof by all parties" is hereby amended to read: "on or before April 12, 2001".

# 2. <u>Miscellaneous</u>.

Except as amended hereby, the Original Agreement remains in full force and effect. This Second Amendment may be executed by facsimile transmission and in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties.

REMAINDER OF PAGE INTENTIONALLY LEFT BLANK IN WITNESS WHEREOF, this Second Amendment has been signed by or on behalf of each of the parties by their duly authorized representatives as of the date first above written.

GATEWAY PIPELINE COMPANY, INC.

By: [

Title: HZESIDFAT

UTILICORP PIPELINE SYSTEMS, INC.

 $\mathbf{B}_{\mathbf{v}}: \mathbf{V}$ 

Title: PRGSIDENT

UTILICORP UNITED INC.

By:

ite: Sr. Vice President

# FIRST AMENDENT TO STOCK PURCHASE AGREEMENT

This First Amendment is dated as of February 16, 2001, by and among Gateway Pipeline Company, Inc., a Delaware corporation ("Purchase"), UtiliCorp United Inc., a Delaware corporation ("Seller"), and UtiliCorp Pipeline Systems, Inc., a Delaware corporation (the "Company").

#### RECITALS

WHEREAS, the parties hereto entered into a Stock Purchase Agreement dated February 1, 2001 (the "Original Agreement"); and

WHEREAS, the parties now desire to amend the Original Agreement.

NOW THEREFORE, in consideration of the mutual promises hereinafter set forth and other good and valuable consideration, the receipt and adequacy of which are herby acknowledged, the parties hereby agree as follows:

#### AGREEMENT

#### 1. Amendment.

HPR.17,2001~10:52HM

- a. Section 1.3. The phrase "on the 11th Business Day" in the first line of said Section is hereby amended to read "on the 21st Business Day".
- b. <u>Section 8.1(c).</u> The phrase "on the 12th Business Day" in the 11<sup>th</sup> line of said Section is hereby amended to read: "on the 22nd Business Day".
- c. <u>Section 8.1(g)</u>. The phrase "within ten Business Days" of said Section is hereby amended to read: "within twenty Business Days".

# 2. Miscellaneous.

Except as amended hereby, the Original Agreement remains in full force and effect. This First Amendment may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties.

K.14.2001, 10.2540, DK144 644

140.663 F.3

IN WITNESS WHEREOF, this First Amendment has been signed by or on behalf of each of the parties by their duly authorized representatives as of the date first above written.

GATEWAY PIPELINE COMPANY, INC.

By:

itle: PRESTOENS

UTILICORP PIPELINE SYSTEMS, INC.

By:

itle: Tersment

UTILICORP UNITED DA

By: \_

Title

#### STOCK PURCHASE AGREEMENT

THIS STOCK PURCHASE AGREEMENT (this "<u>Agreement</u>"), is entered into as of February 1, 2001, by and among Gateway Pipeline Company, Inc., a Delaware corporation ("<u>Purchaser</u>"), UtiliCorp United Inc., a Delaware corporation ("<u>Seller</u>"), and UtiliCorp Pipeline Systems, Inc., a Delaware corporation (the "<u>Company</u>").

#### RECITALS

WHEREAS, Seller owns all of the issued and outstanding shares of capital stock of the Company (the "Shares"); and

WHEREAS, Purchaser desires to purchase and Seller desires to sell the Shares on the terms and conditions of this Agreement; and

WHEREAS, Seller, the Company and Purchaser desire to make certain representations, warranties, covenants and agreements in this Agreement in connection with such purchase and sale;

NOW, THEREFORE, in consideration of the mutual promises hereinafter set forth and other good and valuable consideration, the receipt and adequacy of which are hereby acknowledged, the parties hereby agree as follows:

#### **AGREEMENT**

# ARTICLE I

#### PURCHASE AND SALE OF THE SHARES

- 1.1 Purchase of the Shares from Seller. Subject to and upon the terms and conditions of this Agreement, at the closing of the transactions contemplated by this Agreement (the "Closing") Seller shall sell, transfer, convey, assign and deliver to Purchaser, and Purchaser shall purchase and acquire from Seller, the Shares.
- Subject to adjustment in accordance with Section 1.5, the purchase price to be paid by Purchaser for the Shares shall be the net book value of the property, plant and equipment of the Company and its Subsidiaries (other than Omega Pipeline Company) as of the Closing Date as determined in accordance with GAAP (as defined below) ("Net Book Value") (the "Purchase Price"). The Estimated Adjusted Purchase Price (as determined in accordance with Section 1.5), shall be payable at the Closing to Seller by wire transfer of immediately available funds, to an account to be designated by Seller not less than two Business Days prior to the Closing, or by such other means as the parties may agree. For purposes of this Agreement, "Business Day" shall mean any day on which banks are generally open to conduct business in Missouri and New York, but in any event shall not include Saturdays or Sundays. The Purchase Price is estimated to be \$63.4 million and would be allocated as follows:

Missouri Pipeline - \$32.7 million

Missouri Gas - \$20.4 million

The Company - \$10.3 million

The actual Purchase Price will be allocated among the 3 companies in approximately the same percentages set forth in the above estimates.

- (a) Unless otherwise agreed by the parties hereto prior to the Closing, immediately prior to the Closing, all income tax reserves and pension assets and liabilities on the books of the Company or any Subsidiary (as hereinafter defined) and all intercompany balances (including any cash sweep balances, accounts receivable and accounts payable) between Seller or any Affiliate (as defined hereinafter) of Seller and the Company or any Subsidiary ("Intercompany Balances") shall be distributed, capitalized or otherwise canceled.
- Deposit. On the 11th Business Day after the execution and delivery hereof by all parties (provided Seller has not terminated this Agreement pursuant to Section 8.1(e) hereof), Purchaser shall deliver to Seller by wire transfer of immediately available funds in the amount of \$5,000,000 as an earnest money deposit (the "Deposit"). Pending its disposition as provided herein, the Deposit shall earn interest at an annual rate of 7%. Upon the Closing, the amount of the Deposit (and all interest earned thereon) shall be deducted from the Purchase Price payable under Section 1.2. If Seller terminates this Agreement pursuant to Section 8.1(c) or if the Closing otherwise does not occur due to a breach of this Agreement by Purchaser, then Seller shall be entitled to retain the Deposit (and all interest earned thereon) as its sole and exclusive remedy. If this Agreement is terminated for any reason other than as set forth above, or the Closing otherwise does not occur and Seller is not entitled to retain the Deposit (and all interest earned thereon) pursuant to the above provisions of this Section 1.3, Seller shall immediately return the Deposit (and all interest earned thereon) to Purchaser. Notwithstanding anything in this Agreement to the contrary, should Purchaser for any reason not deliver the Deposit on a timely basis, this Agreement shall terminate as provided in Section 8.1(c) (and Purchaser shall have no liability for failure to deliver the Deposit or the termination of this Agreement).

# 1.4 Closing.

- (a) The Closing shall take place at the offices of Bryan Cave LLP, 3500 One Kansas City Place, 1200 Main Street, Kansas City, MO 64105 commencing at 10:00 a.m., Kansas City time, on the later of (i) June 30, 2001 or (ii) the last Business Day of the month in which all of the conditions to each party's obligations hereunder have been satisfied or waived, or at such other place, time or date as may be mutually agreed upon in writing by the parties (the "Closing Date"). The transfer of the Shares by Seller to Purchaser shall be deemed to occur at 11:59 p.m., Kansas City time, on the Closing Date.
- (b) Subject to the terms and conditions hereof, at the Closing Seller will deliver the following to Purchaser:

- (i) Certificates representing the Shares duly endorsed in blank or with stock powers duly executed in proper form for transfer;
- (ii) The resignations of all members of the Board of Directors and officers of the Company and each Subsidiary; and
- (iii) All other documents, instruments and writings required to be delivered by Seller at or prior to the Closing Date pursuant to this Agreement.
- (c) Subject to the terms and conditions hereof, at the Closing Purchaser will deliver the following to Seller:
  - (i) The Estimated Adjusted Purchase Price as determined in accordance with Section 1.5, less the amount of the Deposit (and all interest earned thereon), in immediately available funds, in the manner set forth in Section 1.2; and
  - (ii) All other documents, instruments and writings required to be delivered by Purchaser at or prior to the Closing Date pursuant to this Agreement.

# 1.5 Working Capital Adjustment.

- (a) The parties understand and acknowledge that the Purchase Price set forth in Section 1.2(a) must be finally determined based on the Net Book Value as of the Closing Date and does not include any amount for Working Capital. The Purchase Price shall be increased or decreased, as the case may be, on a dollar for dollar basis, by the amount of the Working Capital as of the Closing. For purposes of this Agreement, the term "Working Capital" shall be defined as the net amount, positive or negative, of the asset and liability accounts, set forth in <u>Schedule 1.5(a)</u>, the balances of which shall be determined in accordance with Generally Accepted Accounting Principles, applied consistent with past practice ("GAAP").
- (b) At least 10 Business Days prior to the Closing Date, the Company shall deliver to the Purchaser a certificate (an "Estimated Purchase Price Certificate") of the Company signed by an officer of the Company addressed to the Purchaser and prepared by the officer in good faith and on a reasonable basis. The officer shall certify that he has examined the financial and other records of the Company and that, based on such examination, he has determined the estimated amount of the Purchase Price, the Working Capital at the Closing Date and the estimated adjustment to the Purchase Price required by Section 1.5(a). The Estimated Purchase Price Certificate shall state the estimated amount of the Purchase Price, the Working Capital at the Closing Date and such estimated adjustment, and shall set out in reasonable detail the basis of such determination. The Purchase Price (adjusted for the estimated Working Capital) set forth in the Estimated Purchase Price Certificate shall be referred to herein as the "Estimated Adjusted Purchase Price."

- (c) Within 30 days following the Closing, Seller will deliver to Purchaser a certificate (a "Final Purchase Price Certificate") of Seller signed by an officer of Seller addressed to Purchaser and prepared by the officer in good faith and on a reasonable basis. The officer shall certify that he has examined the financial and other records of the Company and that, based on such examination, he has determined the amount of the Purchase Price, and the Working Capital at the Closing Date ("Closing Working Capital") and the adjustment to the Purchase Price required by Section 1.5(a). The Final Purchase Price Certificate shall state the amount of the Purchase Price, the Closing Working Capital, and such adjustment, and shall set out in reasonable detail the basis of such determination.
- (d) Following the delivery of the Final Purchase Price Certificate, the Purchaser shall have a period of 90 days to review the Final Purchase Price Certificate. On or prior to the expiration of such 90-day period, Purchaser shall notify Seller if it accepts such certificate and agrees with the amounts set forth in the Final Purchase Price Certificate or if Purchaser disputes any of such amounts. If the Purchaser does not deliver such notice within such 90-day period, Purchaser shall be deemed to have accepted the Final Purchase Price Certificate as prepared by the Seller. If the Purchaser disputes any of the amounts set forth in the Final Purchase Price Certificate, such notice shall specify the nature of such disagreement in reasonable detail and shall state the opinion of Purchaser as to the correct amount(s).
- (e) Any dispute concerning the Final Purchase Price Certificate will be first referred to appropriate representatives of senior management of each of the parties in written form. If the senior management representatives have not resolved the dispute to the satisfaction of the parties within 20 Business Days after Purchaser's notice under Section 1.5(d) is delivered to the Seller, the matter shall be referred by Seller and Purchaser to Ernst and Young LLP or another mutually agreeable Big 5 accounting firm. The Parties will instruct the accounting firm to proceed promptly to determine the matters in issue and, to the extent possible, to render a decision within 20 Business Days. The accounting firm's decision will be final and binding on the parties and will not be subject to appeal. The fees and expenses of the accounting firm will be apportioned by the accounting firm in accordance with its decision.
- (f) "Final Adjusted Purchase Price" shall mean either (i) the amount of the adjusted Purchase Price set forth in Seller's Final Purchase Price Certificate if such certificate is accepted by Purchaser pursuant to Section 1.5(d), or (ii) the amount of the adjusted Purchase Price determined pursuant to Section 1.5(e). If Final Adjusted Purchase Price exceeds Estimated Adjusted Purchase Price, the Purchaser shall pay to the Seller an amount equal to the amount by which the Final Adjusted Purchase Price exceeds Estimated Adjusted Purchase Price. If Estimated Adjusted Purchase Price exceeds Final Adjusted Purchase Price, Seller shall pay to the Purchaser an amount equal to the amount by which Estimated Adjusted Purchase Price exceeds Final Adjusted Purchase Price. Any amount to be paid by one party to any other party pursuant to the foregoing provisions shall be paid by wire transfer of immediately available funds within 3 Business Days following the determination of Final Adjusted Purchase Price pursuant

to Section 1.5(d) or Section 1.5(e), as the case may be. Such amounts shall bear interest at 7% from the Closing Date until such amount payable under this Section 1.5(f) is paid.

#### ARTICLE II

# REPRESENTATIONS AND WARRANTIES OF SELLER

Seller hereby represents and warrants to Purchaser that:

- 2.1 Corporate Organization. Each of Seller and the Company is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Each of Seller and the Company has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and the Company is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary. True and complete copies of the certificate of incorporation and by-laws of the Company, in effect as of the date of this Agreement, have been delivered to Purchaser.
  - Subject to Section 5.16 below, Schedule 2.1(a) of the Seller's disclosure schedule delivered to Purchaser concurrently herewith (the "Seller Disclosure Schedule") sets forth a complete list of each corporation, partnership, limited liability company, or other organization, whether incorporated or unincorporated, of which the Company owns 50% or more of the outstanding voting securities (each a "Subsidiary" and collectively, the "Subsidiaries") and their respective jurisdictions of incorporation. All of the outstanding shares of capital stock of the Subsidiaries are validly issued, fully paid and non-assessable and such shares are owned by the Company or its wholly-owned Subsidiaries free and clear of any Lien, claim, charge, option, encumbrance, mortgage, pledge, security interest or adverse claim or right of any other party (a "Lien"), restrictions on transfer, or voting or preemptive or other rights with respect thereto. Each Subsidiary (i) is duly incorporated and validly existing and in good standing as a corporation under the laws of its jurisdiction of incorporation, (ii) is duly qualified to do business and in good standing in all jurisdictions (whether federal, state, local or foreign) where its ownership or leasing of property or the conduct of its business requires it to be so qualified, and (iii) has all requisite corporate power and authority to own or lease its properties and assets and to carry on its business as now conducted. As of the date of this Agreement there are not, and at the Closing there will not be, any capital stock or other equity interests in the Subsidiaries issued or outstanding (other than shares of capital stock owned by the Company) or any subscriptions, options, warrants, proxies, voting trusts, calls, rights, convertible securities or other agreements or commitments of any character relating to any stock or other equity interests in any Subsidiary, whether issued or not, or obligating any Subsidiary to issue, transfer or sell any of its capital stock or other equity interests, or any agreements, arrangements, or understandings granting any person any rights in any Subsidiary similar to capital stock or other equity interests. The Company has delivered to Purchaser true and complete copies of the certificates of

incorporation and bylaws of each Subsidiary, in each case as amended to the date of this Agreement.

- 2.2 Capitalization. The authorized capital stock of the Company consists of 1,000 shares of common stock, par value \$1.00 per share, of the Company, of which, as of the date of this Agreement, no shares were held in treasury and only the Shares were issued and outstanding. The Shares have been duly authorized and validly issued and are fully paid, nonassessable and free of preemptive rights, with no personal liability attaching to the ownership thereof. As of the date of this Agreement there are not, and at the Closing there will not be, any capital stock or other equity interests in the Company issued or outstanding (other than the Shares) or any subscriptions, options, warrants, proxies, voting trusts, calls, rights, convertible securities or other agreements or commitments of any character relating to any of the Shares or any stock or other equity interests in the Company, whether issued or not, or obligating the Company to issue, transfer or sell any of its capital stock or other equity interests, or any agreements, arrangements, or understandings granting any person any rights in the Company similar to capital stock or other equity interests.
- 2.3 Authority; No Violation. Each of Seller and the Company has full corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated by this Agreement. The execution and delivery of this Agreement and the consummation of the transactions contemplated by this Agreement have been duly and validly approved by all corporate action on the part of Seller and the Company. No other corporate proceedings on the part of Seller or the Company are necessary to approve this Agreement and to consummate the transactions contemplated by this Agreement. This Agreement has been duly and validly executed and delivered by Seller and the Company and, assuming due authorization, execution and delivery by Purchaser, constitutes a valid and binding obligation of Seller and the Company, enforceable against Seller and the Company in accordance with its terms.
  - (a) Seller has good and marketable title to the Shares, free and clear of any Liens, restrictions on transfer or voting or preemptive or other rights with respect thereto. Seller has the full corporate power, right and authority to transfer, convey and sell the Shares to Purchaser at the Closing.
  - (b) Subject to Section 2.4 and assuming that there are no facts or circumstances concerning Purchaser or any person directly or indirectly controlling, controlled by, or under common control with Purchaser (each such person, an "Affiliate" of Purchaser, with correlative meaning when used with persons other than Purchaser) that require registration or filing with any Governmental Entity (as defined hereinafter) not otherwise contemplated in this Agreement, the execution, delivery and performance of this Agreement by Seller and the Company do not, and the consummation by Seller and the Company of the transactions contemplated by this Agreement will not (i) constitute a breach or violation of, or a default under, the certificate of incorporation or by-laws of Seller, the Company, or any of the Subsidiaries, (ii) conflict with, constitute a breach or violation of, or a default under, or give rise to any Lien, any acceleration of remedies or any right of termination under, any indenture, license, contract, agreement or other

instrument to which the Company or any of the Subsidiaries is a party or by which any of them or their respective properties or assets may be bound, or (iii) assuming compliance with the applicable requirements of the Hart-Scott-Rodino Antitrust Improvements Act of 1976, as amended (the "HSR Act"), violate any Law (as hereinafter defined) of any Governmental Entity applicable to the Company or any of the Subsidiaries or any of their respective properties or assets.

- 2.4 Consents and Approvals. Except as set forth in Schedule 2.4 of the Seller Disclosure Schedule, and except for applicable requirements of the HSR Act, no notice to, filing with, registration, authorization of, exemption by, or consent or approval of, or the taking of any other action in respect of, and no filing, recording, publication or registration with any Governmental Entity or any other person or entity on the part of Seller, the Company or the Subsidiaries in each case is now, or under existing Law of any Governmental Entity in the future will be, necessary for the consummation by Seller, the Company or any of the Subsidiaries of the transactions contemplated by this Agreement.
- 2.5 Licenses; Compliance with Applicable Law. The Company and each of its Subsidiaries hold all licenses, permits, orders, franchises, concessions, approvals, registrations, authorizations and qualifications issued, granted, required or filed by, with or under, as the case may be, any applicable law, ordinance, statute, order, judgment, writ, judicial decision, injunction, decree, award, requirement, rule or regulation ("Law") of any federal, state, political subdivision, commission or other governmental agency or instrumentality, foreign or domestic (each a "Governmental Entity") necessary for the lawful conduct of their respective businesses ("Permits"). There are no proceedings pending or, to Seller's knowledge, threatened which could result in the revocation, cancellation or suspension or any material modification of any Permit and Seller has no knowledge of any facts which if publicly disclosed would be reasonably likely to result in the revocation, cancellation or suspension or any material modification of any Permit. To Seller's knowledge and except as set forth in Schedule 2.5 hereto, the business of the Company and its Subsidiaries is not being and has not been conducted in violation of any Law of any Governmental Entity and the Company and each of its Subsidiaries has complied with and made all filings, including all tariff filings, required pursuant to any Law of any Governmental Entity.
- 2.6 Broker's Fees. Neither Seller nor any of its Subsidiaries nor any of their respective officers or directors has employed any broker or finder or incurred any liability for any broker's fees, commissions or finder's fees in connection with the transactions contemplated by this Agreement.
- 2.7 Financial Statements. Seller has delivered to Purchaser true and complete copies of the unaudited income statements for the Company and each Subsidiary for fiscal year 2000 and balance sheets as of December 31, 2000. All such balance sheets are complete and accurate and fairly present the financial position of the Company and each Subsidiary as of their respective dates, and such income statements fairly present the results of operations for the fiscal year 2000, all in accordance with GAAP. The balance sheet as of December 31, 2000 is referred to herein as the "Balance Sheet". December 31, 2000 is referred to herein as the "Financial Statement Date."

- 2.8 No Undisclosed Liabilities, Claims, etc. The Balance Sheet disclose all liabilities required by GAAP to be disclosed on the Balance Sheet of the Company, except liabilities arising in the ordinary course of business since the Financial Statement Date. To the knowledge of the Seller, the Company has no liabilities of any kind or nature except as set forth on the Balance Sheet, except liabilities arising in the ordinary course of business since the Financial Statement Date. As of Closing, the Company will have no liabilities for indebtedness for borrowed money or advances made from Seller or any of Seller's other subsidiaries.
- Absence of Certain Changes. Except as set forth in Schedule 2.9 of the Seller Disclosure Schedule, since June 30, 2000, there has not been: (a) any adverse change in the business, prospects, financial condition or operations of the Company or any of its Subsidiaries; (b) any damage, destruction or loss, whether covered by insurance or not, adversely affecting the Company's or any Subsidiary's properties and business; (c) any declaration, setting aside or payment of any dividend of stock or property (other than cash) with respect to the Shares, or any redemption or other acquisition of such Shares; (d) any entry by the Company into any commitment or transaction, including, without limitation, any borrowing or capital expenditure, except transactions or commitments entered into in the ordinary course of business of the Company or a Subsidiary; (e) any change in accounting methods, practices or principles as applied to the Company or any of its Subsidiaries, other than changes applicable to Seller generally; (f) any sale, lease or any other disposition or distribution by the Company or any Subsidiary of any of its assets or properties, except transactions in the ordinary course of business of the Company or a Subsidiary or as otherwise consented to by Purchaser; (g) any termination or waiver by Seller, the Company, or any of the Subsidiaries of any rights of value to the business of the Company or any Subsidiary; (h) any conduct by Seller, the Company or any of the Subsidiaries not in the ordinary course of the conduct of the Company's or a Subsidiary's business; or (i) any agreement or understanding made or entered into to do any of the foregoing.
- 2.10 Legal Proceedings. Except as set forth on Schedule 2.10, neither the Company nor any of the Subsidiaries is a party to any, and there are no pending or, to Seller's knowledge, threatened legal, administrative, arbitral or other proceedings, claims, actions or governmental or regulatory investigations of any nature against the Company or any of the Subsidiaries. There are no injunctions, orders, judgments or decrees imposed upon the Company or any of the Subsidiaries.

#### 2.11 Taxes.

- (a) <u>Definitions</u>. For purposes of this Section 2.11 and Article V hereof, the following definitions shall apply:
  - (i) The term "Taxes" shall be deemed to include without limitation all federal, possession, state, city, county and foreign (or governmental unit, agency, or political subdivision of any of the foregoing) income, profits, employment (including Social Security, unemployment insurance and employee income tax withholding), franchise, gross receipts, sales, use, transfer, stamp, occupation, property, ad valorem and excise taxes; Pension Benefit Guaranty Corporation premiums and any other governmental charges of the same or similar

nature; and all penalties, additions to tax and interest relating to any such taxes, premiums or charges. Any one of the foregoing Taxes shall be referred to sometimes as a "Tax."

- (ii) The term "Returns" shall be deemed to include, without limitation, all reports, estimates, information statements and returns (including information returns and reports (e.g., Forms W-2 and Forms 1099)) relating to or required by Law to be filed by the Company in connection with any Taxes and, in the case of consolidated or combined tax returns, by Seller on behalf of the Company. Any one of the foregoing Returns shall be referred to sometimes as a "Return."
- (iii) The term "Tax Affiliate" shall mean, with respect to a corporation, any member of an "Affiliated group" as defined in section 1504 of the Code, or member of a combined or unitary group of which such corporation is or was a member (other than such corporation).
- (iv) The term "Code" shall refer to the Internal Revenue Code of 1986, as amended, and/or, where appropriate, its predecessor, the Internal Revenue Code of 1954, as amended.
- (b) Member of Affiliated Group. Since its formation, the Company and each of its Subsidiaries have been a member of an "Affiliated group" of corporations within the meaning of section 1504 of the Code, with respect to which Seller is and at all times has been the common parent, and has joined in or will join in the filing of Seller's consolidated federal income tax returns for all its taxable periods ending on or prior to the Closing Date.
- (c) Returns Filed and Taxes Paid. All Returns required to be filed by or with respect to the Company and each of its Subsidiaries have been filed on a timely basis and all such Returns are complete and accurate. All Taxes due and owing with respect to the Company and its Subsidiaries prior to the Closing whether or not reflected in the Returns as filed, have been paid in full or adequately reserved for on the Closing Date balance sheet (and thus included in the Working Capital Adjustment).
- (d) Withholding Taxes. Company has complied with all applicable laws, rules, and regulations relating to the withholding of Taxes and the payment thereof (including, without limitation, withholding of Taxes under Section 1441 and 1442 of the Code, or similar provision under foreign laws), and timely and properly withheld from employee wages and paid over to the proper governmental authorities all amounts required to be withheld and paid over under applicable laws.
- (e) Section 338(h)(10) Representations. Seller has filed a consolidated federal income tax return with the Company for the taxable year immediately preceding the current taxable year, and Seller is eligible to make an election under section

338(h)(10) of the Code (and any comparable election under state, local or foreign tax law) with respect to the Company.

## 2.12 Contracts.

- (a) The Company has delivered to Purchaser at the offices of the Company true and complete copies of all contracts and agreements of the following categories to which the Company or any of its Subsidiaries is a party or by which any of their respective properties or assets are bound (the "Company Contracts"):
  - (i) material license, distributor, dealer, sales agency, advertising, property management or brokerage agreements;
  - (ii) agreements for the future purchase of more than \$10,000 of materials, supplies, services, merchandise or equipment;
  - (iii) agreements under which the Company or any Subsidiary interconnects with any other interstate, intrastate or local distribution pipeline or pipelines (i.e. interconnect or balancing agreements);
  - (iv) agreements under which the Company or any Subsidiary is obligated to: (A) gather, treat, compress or transport natural gas, or undertake such gathering, treating, compressing or transportation, for a third party; or (B) process or have processed natural gas (a "Gas Transportation Contract");
  - (v) agreements for the sale or lease of the Company's or any Subsidiary's property or assets, or the grant of any preferential right to purchase any of the Company's or any Subsidiary's property or assets or requiring the consent of any party to the transfer and assignment of the Company's or any Subsidiary's property or assets, other than those entered into in the ordinary course of the Company's or any Subsidiary's business;
  - (vi) agreements, indentures or other instruments relating to the
     borrowing of money or the guarantee of any obligation, other than such agreements, indentures and other investments that will be terminated on or before Closing;
    - (vii) franchises, licenses, royalty or other Agreements respecting or restricting any Intellectual Property Rights (as hereinafter defined);
    - (viii) agreements made other than in the ordinary course of the Company's or any Subsidiary's business;
    - (ix) agreements with any Affiliate of the Company, except Agreements among only the Company and any of its Subsidiaries; or

(x) agreements limiting the freedom of the Company or any Subsidiary to compete in any line of business or with any party.

# Schedule 2.12(a) of the Seller Disclosure Schedule lists all of the Company Contracts.

- (b) Neither the Company nor any of its Subsidiaries has any contracts whereby they are obligated to purchase any natural gas from or sell natural gas to any person.
- (c) There is no default under any Company Contract by the Company or any Subsidiary or, to Seller's knowledge, by any other party thereto, and no event has occurred or condition exists that with the lapse of time or the giving of notice or both would constitute a default thereunder by the Company or any Subsidiary or, to Seller's knowledge, any other party. No party to any Company Contract has given notice to the Company or any Subsidiary of or made a claim against the Company or any Subsidiary with respect to any breach or default thereunder.
- (d) <u>Schedule 2.12(d)</u> of the Seller Disclosure Schedule sets forth a true and correct summary of the term and price provisions of each Gas Transportation Contract.
- (e) <u>Schedule 2.12(e)</u> of the Seller Disclosure Schedule sets forth all unresolved disputes under any Gas Transportation Contract.
- (f) Since December 31, 1998, there has been no transportation of natural gas by the Company or any Subsidiary in interstate commerce that would invoke or subject the Company or any Subsidiary or their respective assets or properties to the jurisdiction of the Federal Energy Regulatory Commission ("FERC") under the Natural Gas Act of 1938, including all amendments thereto.
- Assets. The Company and/or the Subsidiaries have good and marketable title to all of the properties and assets reflected in the Balance Sheet or acquired after the date thereof (except properties sold or otherwise disposed of since the date thereof in the ordinary course of business) free and clear of all Liens, except for Permitted Encumbrances. Schedule 2.13A contains a true and accurate list of all vehicles owned by the Company. As used herein, the term "Permitted Encumbrances" shall mean, with respect to or upon any of the property or assets of the Company or its Subsidiaries, whether owned as of the date hereof or thereafter, any (1) Liens on property of the Company and its Subsidiaries existing on the date of this Agreement and set forth on Schedule 2.13B of the Seller Disclosure Schedule; (2) Liens incurred and pledges and deposits made in the ordinary course of business in connection with workers' compensation; (3) Liens imposed by law, such as carriers', warehousemen's, mechanics', materialmen's, landlords', laborers', suppliers' and vendors' liens, incurred in good faith in the ordinary course of business and securing obligations which are not yet due or which are being contested in good faith by appropriate proceedings as to which the Company or such Subsidiary, as the case may be, shall, to the extent required by GAAP, have set aside on its books adequate reserves; (4) Liens securing the payment of Taxes, either not delinquent or being contested in

good faith by appropriate legal or administrative proceedings and as to which the Company or such Subsidiary, as the case may be, shall, to the extent required by GAAP, have set aside on its books adequate reserves; (5) with respect to owned real property, zoning restrictions and easements, licenses, rights of way, declarations, reservations, provisions, covenants, conditions, waivers, restrictions on the use of property or other title matters of record (and with respect to leasehold interests, mortgages, obligations and Liens incurred, created, assumed or permitted to exist and arising by, through or under a landlord or owner of the leased property, with or without consent of the lessee); and (6) extensions, renewals and replacements of Liens referred to in (1) through (5) of this sentence (provided, that any such extension, renewal or replacement Lien shall be limited to the property or assets covered by the Lien extended, renewed or replaced and that the obligations secured by any such extension, renewal or replacement Lien shall be in an amount not greater than the amount of the obligations secured by the original Lien extended, renewed or replaced); provided, that in each case referred to in (1) through (6) above, such Liens would not materially impair or prevent the operation, value or use of the Company's or its Subsidiaries' assets as presently used, and do not present financial obligations or liabilities of the Company or its Subsidiaries. The Company and the Subsidiaries own or lease, directly or indirectly, all of the assets and properties necessary to carry on their respective businesses or operations as presently conducted. No assets owned by Omega (as defined in Section 5.18 below) are used in the operation of Company or its other Subsidiaries. TransMississippi Pipeline Company ("TMP") was merged with and into the Company as of June 30, 1999.

# (a) Real Property.

- (i) Schedule 2.13(a)(i) of the Seller Disclosure Schedule contains a list of all leases of real property under which the Company or any Subsidiary is a tenant. With respect to the leased real property described on Schedule 2.13(a)(i) of the Seller Disclosure Schedule, to Seller's knowledge (A) all such leases are in full force and effect and constitute valid and binding obligations of the respective parties thereto; (B) there have not been and there currently are not any defaults thereunder by any party thereto; (C) no event has occurred which (whether with or without notice, lapse of time or the happening or occurrence of any other event) would constitute a default thereunder entitling the lessor to terminate the lease; and (D) the continuation, validity and effectiveness of all such leases under the current rentals and other current terms thereof will in no way be affected by the transactions contemplated by this Agreement.
  - (ii) Schedule 2.13(a)(ii) of the Seller Disclosure Schedule contains a map showing the pipeline of the Company and its Subsidiaries and a list of all easements or rights of way, road crossings, railroad crossings, water body crossings and any other real property rights (the "Easements"), reasonably necessary to carry on the respective business or operations of the Company or any Subsidiary as presently conducted as shown on said map, it being the intention of the parties that by acquiring the Shares, Purchaser is acquiring the pipeline and the Easements (through the ownership thereof by the Company and its Subsidiaries).

- (iii) Each parcel of real property, building, structure and improvement owned by the Company or any Subsidiary (collectively the "Premises") conforms to all applicable Laws of any Governmental Entity in all material respects, including zoning regulations, none of which will, upon the sale of the Shares to Purchaser, prohibit the use of such properties, buildings, structures or improvements, for the purposes for which they are now utilized.
- (b) <u>Personal Property</u>. The Company and the Subsidiaries hold fee and/or leasehold title to all the personal property and assets, tangible or intangible, shown on the Balance Sheet of a quality sufficient to allow Purchaser to conduct the business of the Company and the Subsidiaries as presently conducted. None of such assets are subject to any Liens except for Permitted Encumbrances and standard equipment leases.
- (c) <u>Accounts Receivable</u>. The accounts receivable of the Company and each Subsidiary reflected in the Balance Sheet (the "<u>Accounts</u>") represent valid obligations arising from sales actually made or services actually performed in the ordinary course of the Company's and each Subsidiary's business. There is no contest, claim or right of set-off under any Company Contract with any obligor of an Account relating to the amount or validity of such Account.
- 2.14 Reports and Inspections. Schedule 2.14 of the Seller Disclosure Schedule contains a list of all material reports of inspections by representatives of any Governmental Entity of the business and properties of the Company or any Subsidiary from January 1, 1999 through the date hereof under OSHA and under all other applicable health and safety Laws of any Governmental Entity.
- 2.15 Absence of Certain Business Practices. To Seller's knowledge, none of Seller, the Company, any of the Subsidiaries, or any other Affiliate of Seller, acting alone or together, has, directly or indirectly: (a) received any rebates, payments, commissions, promotional allowances or any other economic benefit, regardless of its nature or type, from any customer, supplier, trading company, shipping company, governmental employee or other entity or individual with whom the Company or any Subsidiary has done business directly or indirectly; or (b) given or agreed to give any gift or similar benefit to any customer, supplier, trading company, shipping company, governmental employee or other person or entity who is or may be in a position to help or hinder the business of the Company or any Subsidiary (or assist the Company or any Subsidiary in connection with any actual or proposed transaction) which, in either case (a) or (b), would be reasonably expected to subject the Company or any Subsidiary to any damage or penalty in any criminal or governmental litigation or proceeding.

#### 2.16 Environmental Matters.

- (a) The following definitions shall be applicable only to this Section 2.16:
  - (i) "Environmental Laws" shall mean any applicable laws, statutes, regulations, rules, ordinances, and codes of governmental agencies of the

United States, the states and localities or municipalities currently in effect designed to minimize, prevent, punish, or remedy damages or threats of damage to the environment or public health caused by the manufacture, use, treatment, storage, disposal, or handling of Hazardous Materials.

- (ii) "Hazardous Materials" shall mean chemicals, materials, substances or wastes which are listed, designated, classified as, defined or otherwise determined to be, "hazardous substances", "hazardous materials", "toxic substances", "extremely hazardous substances", or words of similar import or otherwise hazardous or toxic or a pollutant or contaminant under or pursuant to any Environmental Law and specifically including petroleum and friable asbestos or friable asbestos containing materials.
- (iii) "Release" shall mean any spilling, emitting, discharging, leaking, pouring, emptying, leaching, escaping, dumping, disposing or other releasing of Hazardous Materials into the environment, whether intentional or unintentional.

# (iv) "Environmental Claim" shall mean any:

- (A) fines, penalties, judgments, awards, settlements, legal or administrative proceedings, damages, losses, claims, demands and response, investigative, remedial, or inspection costs and expenses arising under Environmental Laws; and
- (B) any pending or threatened administrative, regulatory or judicial actions or causes of action, suits, judgments, orders, and notices of noncompliance or violation relating to any Environmental Laws or arising from the presence or Release of any Hazardous Materials.
- (v) "Facilities" shall mean any real property currently owned, leased, or operated by the Company or its Subsidiaries and any buildings, structures, or equipment currently owned or operated by the Company or its Subsidiaries.
- (b) Except as set forth in <u>Schedule 2.16(b)</u> of the Seller Disclosure Schedule:
  - (i) the Company and its Subsidiaries are in compliance with all applicable Environmental Laws;
  - (ii) Seller, the Company and its Subsidiaries have not received any written notice alleging that the Company or its Subsidiaries may be in violation of or subject to any liability under any Environmental Law;
  - (iii) Seller, the Company and it Subsidiaries have not directly or proximately caused any Release of Hazardous Materials at any Facilities or any

other locations controlled by the Company or its Subsidiaries in which the Company or its Subsidiaries has or had an interest which would require reporting or require remedial action under or pursuant to Environmental Law;

- (iv) there are no pending or, to Seller's knowledge, threatened Environmental Claims with respect to or adversely affecting any of the Facilities or any other properties and assets in which the Company or its Subsidiaries has or had an interest;
- (v) Seller has delivered to Purchaser true and complete copies and results of any reports, studies, analyses, tests, or monitoring possessed or initiated by the Company or its Subsidiaries pertaining to the Release of Hazardous Materials in, on, or under the Facilities, or concerning compliance by the Company or its Subsidiaries with Environmental Laws;
- (vi) to Seller's knowledge, except for materials that are commonly present or utilized in the ordinary course of the businesses of the Company and the Subsidiaries, there are no Hazardous Materials present on or at the Facilities, including any Hazardous Materials contained in barrels, above or underground storage tanks, landfills, land deposits, dumps, equipment (whether moveable or fixed) or other containers, either temporary or permanent, and deposited or located in land, water, sumps or any other part of the Facilities or incorporated into any structure therein or thereon; and
- (vii) Seller, the Company and its Subsidiaries have not installed any underground storage tanks at any of the Facilities and, to Seller's knowledge, there are no underground storage tanks at any of the Facilities.
- 2.17 Intellectual Property. Except for the Trademarks and Logos (as hereinafter defined), the Company or a Subsidiary owns or licenses (and accordingly has the right to use), and Schedule 2.17 of the Seller Disclosure Schedule contains a listing of, each copyright, trademark, trade name, service mark, trade dress, patent, registration of any of the foregoing, and application for any patent or registration (collectively "Intellectual Property Rights") necessary for the Company and the Subsidiaries to carry on their respective businesses as currently To Seller's knowledge, the conduct of the business of the Company and its Subsidiaries (including the use of the Intellectual Property Rights) as currently conducted does not infringe upon the proprietary rights of any third party and there are no present or threatened infringements of the Intellectual Property Rights by any third party. There are no pending or, to Seller's knowledge, threatened proceedings or litigation or other adverse claims by any person against the use by the Company or any of its Subsidiaries of any Intellectual Property Rights. Except as otherwise set forth on Schedule 2.17 of the Seller Disclosure Schedule, all of said Intellectual Property Rights, the right to use them and the right to convey them are free and clear of all Liens and royalty obligations.
- 2.18 Employee Benefits. Except as set forth on Schedule 2.18 of the Seller Disclosure Schedule, neither the Company nor any Subsidiary maintains, contributes or is

required to contribute to any "employee welfare benefit plan" (within the meaning of Section 3(1) of the Employee Retirement Income Security Act of 1974 ("ERISA")), any multiemployer plan, any "employee pension benefit plan" (within the meaning of Section 3(2) of ERISA) or any fringe benefit, profit sharing, deferred compensation, bonus, stock option, stock purchase, pension, retainer, consulting, retirement, welfare, or other incentive plan or Agreement (collectively, "Employee Benefit Plans") and has no liability (whether known or unknown, absolute, accrued, fixed or contingent, matured or unmatured, or otherwise) for any Employee Benefit Plan maintained or contributed to by any Affiliate of the Company or any Subsidiary.

- 2.19 Employees. Neither the Company nor any Subsidiary has any employees. All persons engaged primarily in the business and operations of the Company and each Subsidiary are listed on <u>Schedule 2.19</u> of the Seller Disclosure Schedule, which for each such person lists the current job title and base salary.
- 2.20 Insurance. Seller has insurance policies (and amounts and types of coverage) that are, in the aggregate, sufficient to protect and insure the Company and each Subsidiary against perils which good business practice demands be insured against or which are normally insured against by other industry members similarly situated.
- 2.21 Suppliers. Since the Financial Statement Date, no supplier of goods or services to the Company or any Subsidiary that has since the Financial Statement Date made sales or provided services representing more than \$25,000 in payments or commitments by the Company or any Subsidiary has: (a) ceased, or indicated any intention to cease, doing business with the Company or any Subsidiary; or (b) changed or indicated any intention to change any material terms or conditions for future supply or sale of products or services.
- 2.22 Bank Accounts. Schedule 2.22 of the Seller Disclosure Schedule lists all bank accounts, lock boxes, post office boxes and safe deposit boxes maintained in the name of or controlled by the Company or any Subsidiary and the names of the persons having access thereto.
- 2.23 Books and Records and Agreements. The books of account, minute books, stock record books and other records of the Company and the Subsidiaries, all of which have been either delivered to Purchaser or made available to Purchaser in accordance with the terms of this Agreement, are complete and correct in all material respects.
- 2.24 Public Utility Holding Company Act. None of the Company or any of its Subsidiaries is a "public utility company" or a "holding company" or a "subsidiary company" of a "holding company" within the meaning of the Public Utility Holding Company Act of 1935, as amended.
- 2.25 Regulatory. Except as disclosed on Schedule 2.25 of the Seller Disclosure, (a) Missouri Gas Company ("MGC") and Missouri Pipeline Company ("MPC") each hold all required certificates of public convenience and necessity to conduct their respective businesses as presently conducted and are in full compliance with all of the terms and conditions of such certificates, (b) MGC and MPC have no current requirements under any prior or current order, directive, demand or request to have to file for any review or other determination of its

rates, tariffs and/or operating terms and conditions and (c) none of Seller, the Company, MGC or MPC have received any written or verbal communication from any governmental or regulatory body directing, demanding or requesting that MPC or MGC file for any review or other determination of its rates, tariffs and/or operating terms and conditions.

- 2.26 Customers. The 5 largest customers of the Company in the last fiscal year and the percentage of the gross revenue of the Company contributed by each is set forth on Schedule 2.26 hereto (the "Material Customers"). To the best of Seller's and the Company's knowledge, no Material Customer (i) has cancelled or threatened to cancel or indicated its intention to cancel or otherwise materially adversely modify its relationship with the Company, (ii) is in breach of any contract with the Company (iii) has asserted that any contract with the Company (or any material provision thereof, including the rates set forth therein) is invalid or unenforceable or (iv) has asserted that it has a right of set off for any reason of any amount in excess of \$5,000.
- 2.27 Disclosure. No representation or warranty made in this Agreement contains any untrue statement of a material fact necessary to make the statement made therein, in light of the statements made therein, not misleading.

## ARTICLE III

## REPRESENTATIONS AND WARRANTIES OF PURCHASER

Purchaser hereby represents and warrants to Seller that, except as disclosed in the Purchaser's disclosure schedule delivered to Seller concurrently herewith (the "<u>Purchaser Disclosure Schedule</u>"):

- 3.1 Corporate, Organization. Purchaser is a corporation duly organized, validly existing and in good standing under the laws of the State of Delaware. Purchaser has the corporate power and authority to own or lease all of its properties and assets and to carry on its business as it is now being conducted, and is duly licensed or qualified to do business in each jurisdiction in which the nature of the business conducted by it or the character or location of the properties and assets owned or leased by it makes such licensing or qualification necessary, except where the failure to be so licensed or qualified would not adversely affect the ability of Purchaser to consummate the transactions contemplated by this Agreement. True and complete copies of the articles of incorporation and by-laws of Purchaser, as in effect as of the date of this Agreement, have previously been made available by Purchaser to Seller.
- 3.2 Authority; No Violation. Purchaser has full corporate power and authority to execute and deliver this Agreement and to consummate the transactions contemplated by this Agreement. The execution and delivery of this Agreement and the consummation of the transactions contemplated by this Agreement have been duly and validly approved by the Board of Directors of Purchaser. No other corporate proceedings on the part of Purchaser are necessary to approve this Agreement and to consummate the transactions contemplated by this Agreement. This Agreement has been duly and validly executed and delivered by Purchaser and, assuming

due authorization, execution and delivery by Seller and the Company, constitutes a valid and binding obligation of Purchaser, enforceable against Purchaser in accordance with its terms.

- Assuming that there are no facts or circumstances concerning Seller, the Company or their respective Affiliates that require registration or filing with any Governmental Entity not otherwise contemplated in this Agreement, the execution. delivery and performance of this Agreement by Purchaser do not, and the consummation by Purchaser of the transactions contemplated by this Agreement will not, (i) constitute a breach or violation of, or default under the articles of incorporation or by-laws of Purchaser, (ii) conflict with, constitute a breach or violation of, or a default under, or give rise to any Lien, any acceleration of remedies or any right of termination under, any indenture, license, contract, agreement or other instrument to which Purchaser or of any of its Subsidiaries is a party or by which any of them or their respective properties or assets may be bound, or (iii) assuming compliance with the applicable requirements of the HSR Act, violate any Law of any Governmental Entity applicable to Purchaser or any of its Subsidiaries or any of their respective properties or assets (but no representation is given regarding any approvals by the Missouri Public Service Commission), except in the case of (ii) and (iii) above for such breaches, violations, defaults, liens, accelerations or rights as would not be reasonably expected, individually or in the aggregate, to adversely affect the ability of Purchaser to consummate the transactions contemplated by this Agreement.
- 3.3 Consents and Approvals. Except for the applicable requirements of the HSR Act, no notice to, filing with, registration, authorization of, exemption by, or consent or approval of, or the taking of any other action in respect of any Governmental Entity or any other entity on the part of Purchaser is necessary for consummation by Purchaser or any of its Subsidiaries of the transactions contemplated by this Agreement, except where the failure to provide such notice, make such filing, or obtain such authorization, exemption, conflict or approval would not, individually or in the aggregate, be reasonably expected to adversely affect the ability of Purchaser to consummate the transactions contemplated by this Agreement.
- 3.4 Broker's Fees. Neither Purchaser nor any of its Affiliates nor any of their respective officers or directors has employed any broker or finder or incurred any liability for any broker's fees, commissions or finder's fees in connection with the transactions contemplated by this Agreement.
- 3.5 Legal Proceedings. Neither Purchaser nor any of its Affiliates is a party to any, and there are no pending or, to Purchaser's knowledge, threatened legal, administrative, arbitral or other proceedings, claims, actions, or governmental or regulatory investigations of any nature against Purchaser or any of its Affiliates which, individually or in the aggregate, would reasonably be expected adversely affect the ability of Purchaser to consummate the transactions contemplated by this Agreement.
- 3.6 Acquisition of Shares for Investment. Purchaser has such knowledge and experience in financial and business matters that it is capable of evaluating the merits and risks of its purchase of the Shares. Purchaser is acquiring the Shares for investment and not with a view

toward or for sale in connection with any distribution thereof, or with any present intention of distributing or selling the Shares. Purchaser agrees that the Shares may not be sold, transferred, offered for sale, pledged, hypothecated or otherwise disposed of without registration under the Securities Act of 1933, as amended, except pursuant to an exemption from such registration available under such Act, and without compliance with foreign securities laws, in each case, to the extent applicable.

## ARTICLE IV

# COVENANTS OF THE PARTIES

- Investigation of Business; Access to Properties and Records. From the date of this Agreement through the Closing, Seller shall use commercially reasonable efforts to cause the Company and its Subsidiaries to afford to representatives of Purchaser reasonable access to the offices, properties, employees, customers, suppliers, books and records of the Company and its Subsidiaries during normal business hours, in order that Purchaser may have a full opportunity to make such investigations as it desires of the affairs of the Company and its Subsidiaries, provided, however, that such investigation shall be at reasonable times and upon reasonable notice and shall not unreasonably disrupt the personnel and operations of any of Seller, the Company or any of the Subsidiaries. All requests for access to the offices, properties, books, and records relating to the Company or its Subsidiaries shall be made to such representatives of Seller as Seller shall designate, who shall be solely responsible for coordinating all such requests and all access permitted hereunder. Without limiting Seller's obligations set forth in the first sentence of this Section, neither Purchaser nor its representatives shall contact any of the employees, customers or suppliers of any of Seller, the Company or the Subsidiaries, in connection with the transactions contemplated by this Agreement, whether in person or by telephone, mail or other means of communication, except as may be jointly scheduled by Purchaser and Seller.
  - (a) Any information provided to Purchaser or its representatives pursuant to this Agreement shall be held by Purchaser and its representatives in accordance with, and shall be subject to the terms of, the confidentiality agreement dated January 10, 2000 entered into between Seller and R2 Development Inc., (the "Confidentiality Agreement").
  - (b) Following the Closing Date, Purchaser shall provide Seller, its accountants and counsel, during normal business hours, upon reasonable notice, full access to the books, records, properties and employees of the Company and the Subsidiaries to the extent that such access may be requested for any legitimate purpose at no cost to Seller (other than for reasonable out-of-pocket expenses).
- 4.2 Consents and Approvals. Subject to the terms and conditions of this Agreement, Purchaser and Seller each agree to use their respective reasonable best efforts to promptly (i) take, or cause to be taken, all actions and to do, or cause to be done, all things necessary, proper or advisable under applicable Laws of any Governmental Entity to consummate and make effective the transactions contemplated by this Agreement, (ii) obtain and maintain all

approvals, consents, registrations, permits, authorizations and other confirmations required to be obtained from any third party that are necessary, proper or advisable to consummate the transactions contemplated by this Agreement, (iii) lift or rescind any injunction or restraining order or other order adversely affecting its ability to consummate the transactions contemplated by this Agreement, and (iv) fulfill all conditions to this Agreement. In no event, however, will the Company or any Subsidiary, prior to the Closing, commit to any divestiture, or similar transaction or action with respect to any asset or business of the Company or its Subsidiaries, which commitment and implementation may be conditioned upon and effective as of the Closing, for the purpose of obtaining any such approval, consent, registration, permit, authorization or other confirmation without Purchaser's prior written consent. Furthermore, in no event shall either Seller, Purchaser, the Company or any Subsidiary be obligated to (x) pay any material amount of money to any person or to offer or grant other financial or other accommodations to any person in connection with its obligations under this Section 4.2 or (y) provide any undertaking or agreement, or take or refrain from taking any other action in connection with either party's obligations under this Section 4.2, which, in such party's reasonable judgment, would, individually or in the aggregate, adversely affect the continuing business of such party or any of its Affiliates. Subject to applicable Laws of any Governmental Entity relating to the exchange of information, Seller and Purchaser shall have the right to review, in advance, and to the extent practicable each will consult the other on, all submissions and communications relating to the Company and the Subsidiaries or Purchaser, as the case may be, made with any third party and/or any Governmental Entity in connection with the transactions contemplated by this Agreement.

- (a) In furtherance and not in limitation of the foregoing, each of Purchaser and Seller agrees to (i) make an appropriate filing of a Notification and Report Form pursuant to the HSR Act with respect to the transactions contemplated by this Agreement as promptly as practicable and in any event within thirty Business Days of the date of this Agreement, (ii) supply as promptly as practicable any additional information and documentary material that may be requested pursuant to the HSR Act and (iii) use reasonable commercial efforts (not including the expenditure of amounts in excess of \$100,000) to complete the review process under the HSR Act to permit the consummation of the transactions contemplated by this Agreement.
- (b) In furtherance and not in limitation of the foregoing, if the parties determine in their reasonable discretion that a party must (i) obtain from any Governmental Entity consents, waivers, approvals, authorizations, or Permits in connection with the authorization, execution, and delivery of this Agreement or the consummation of the transactions contemplated hereby or (ii) make any applications or filings with a Governmental Entity under any applicable public utility laws or regulations or any other applicable law of any Governmental Entity with respect to this Agreement or the transactions contemplated hereby, including without limitation applications for approval by any Governmental Entity or a request for a declaratory ruling from any Governmental Entity with respect to the jurisdiction of such Governmental Entity over the transactions contemplated hereby, the parties shall fully cooperate with each other in such regard and use their respective reasonable best efforts to obtain such Permits and to make such applications or filings.

- 4.3 Further Assurances. Seller and Purchaser agree that, from time to time, whether before, at or after the Closing Date, each of them will execute and deliver such further instruments of conveyance and transfer and take such other action as may be necessary to carry out the purposes and intents of this Agreement.
- 4.4 Conduct of Business. Without limiting the Seller's obligations pursuant to Section 4.5 hereof, from the date of this Agreement through the Closing, except as disclosed on Schedule 4.4 attached hereto or otherwise provided for in, or contemplated by, this Agreement, and except as consented to or approved by Purchaser in writing (which consent or approval shall not be unreasonably withheld), Seller covenants and agrees that:
  - (a) each of the Company and the Subsidiaries shall operate its business in the ordinary course in all material respects;
  - (b) none of the Company or the Subsidiaries shall (i) amend its respective certificate or articles of incorporation or by-laws or comparable organizational documents or (ii) make any change in its authorized or issued capital stock;
  - (c) none of the Company or the Subsidiaries shall (i) issue, sell or agree to issue or sell any shares of its capital stock, or any securities convertible into, or options with respect to, or warrants to purchase or rights to subscribe for, any shares of its capital stock or (ii) redeem, purchase or otherwise acquire, directly or indirectly, any of the Shares or other securities of the Company or any Subsidiary; and
  - (d) none of Seller, the Company or the Subsidiaries will, or will cause the Company or any Subsidiary to, take any affirmative action, or fail to take any reasonable action within their control, as a result of which any of the changes or events listed in Section 2.9 is likely to occur.

Without limiting the rest of Section 4.4, from the date of this Agreement through the Closing, none of the Company or the Subsidiaries shall enter into any agreement (or renewal of any contracts with existing customers) (i) with a term extending beyond December 31, 2001, other than gas transportation agreements at maximum tariff rates, or (ii) which requires construction of any new facilities or purchase of any new equipment.

Notwithstanding the provisions of this Section 4.4, nothing in this Agreement shall be construed or interpreted to prevent the Company or the Subsidiaries from (i) paying or making regular or special dividends or other distributions consisting of cash, (ii) making or accepting inter- or intra-company advances to, from or with one another or with Seller or any of its Affiliates, or (iii) engaging in any transaction incident to the normal cash management procedures of Seller and its Affiliates.

4.5 Preservation of Business. Subject to the terms and conditions of this Agreement, Seller shall, and shall cause the Company and the Subsidiaries to, use reasonable commercial efforts to preserve the Company's business intact, to keep available to the Company and the Subsidiaries the services of the Company Employees (as defined below) and to preserve

the goodwill of customers and others having business relations with the Company and the Subsidiaries in all material respects.

- 4.6 Public Announcements. Except as otherwise required by law, each of Seller and Purchaser will consult with the other and obtain the consent of the other (which consent shall not be unreasonably withheld or delayed) before issuing, or permitting any agent or Affiliate to issue, any press releases or otherwise making or permitting any agent or Affiliate to make, any public statements with respect to this Agreement and the transactions contemplated by this Agreement; provided, however, the parties agree that the Seller may make such disclosure as may be required by applicable securities law or by obligations pursuant to any listing agreement with any national securities exchange without prior consultation. No press release or public statement with respect to the Agreement or the transactions contemplated hereby shall be made until Purchaser delivers the Deposit to Seller under Section 1.3 above subject to requirement of applicable law (including securities laws) and such listing agreements.
- Corporate Names. Purchaser acknowledges that, from and after the Closing Date, Purchaser shall have no rights with respect to any names, marks, trade names, trademarks and logos (collectively, "Trademarks and Logos") incorporating "UtiliCorp" or "EnergyOne" or "Aquila" by themselves or in combination with any other Trademark or Logo, including, without limitation, the corporate design logos associated therewith, and that Seller shall retain absolute and exclusive proprietary rights thereto or goodwill represented thereby or pertaining thereto. Purchaser agrees that promptly after the Closing Date it will amend the Company's certificate of incorporation to remove the "UtiliCorp" or "EnergyOne" or "Aquila" name and that from and after the Closing Date it will not, nor will it permit any of its Affiliates to, use any name, phrase or logo incorporating "UtiliCorp" or "EnergyOne" or "Aquila" or such corporate design logo or any confusingly similar name, phrase, logo or corporate design logo in or on any of its literature, sales materials or products or otherwise in connection with the sale of any products or services; provided however, that Purchaser may continue to use any printed literature, sales materials, purchase orders and sales or lease agreements, and sell any products, that are included in the inventories of the Company or any of its Subsidiaries on the Closing Date and that bear a name, phrase or logo incorporating "UtiliCorp" or "EnergyOne" or "Aquila" or such corporate design logo, until the supplies thereof existing on the Closing Date have been exhausted, but in any event for not longer than 30 days from the Closing Date or, in the case of field location signs, 180 days from the Closing Date.
- 4.8 Insurance. Purchaser acknowledges that the programs and policies of insurance maintained by Seller to provide coverage in favor of the Company and the Subsidiaries will be terminated effective 12:01 A.M. on the day following the Closing Date. Except as contemplated in this Agreement, from and after the Closing Date, all risk of loss with respect to properties and assets of the Company and the Subsidiaries shall be borne by Purchaser.
- 4.9 Guaranties. Purchaser shall use its reasonable commercial efforts (including an offer of a substitute guarantee, letter of credit or undertaking of the Purchaser) to cause or procure the release, as of the Closing Date, of all liabilities or obligations of Seller or any Affiliate of Seller (other than the Company and Subsidiaries) (each of Seller and any such Affiliate other than the Company and the Subsidiaries being referred to in this Section 4.9 as a

"Guarantor") from any guarantees (whether of payment or performance), letters of credit or other similar undertakings or credit support arrangements such Guarantor has delivered to others for the benefit of the Company or any Subsidiary, all of which are listed on Schedule 4.9 hereto. If after the Closing Date one or more Guarantor(s) remains liable for any such liabilities, Purchaser shall continue to use its reasonable commercial efforts to cause or procure the Guarantors' release from such liabilities and agrees to indemnify and hold harmless such Guarantor(s) from any cost, expense or loss (including reasonable attorneys' fees) incurred by such Guarantor(s) arising directly or indirectly therefrom. Any reasonable out-of-pocket expense incurred by any Guarantor with respect to maintaining such support for periods after the Closing Date shall be reimbursed to such Guarantor by the Company or Purchaser.

Notification. Between the date hereof and the Closing Date, each of Seller 4.10 and Purchaser will promptly notify the other in writing if it becomes aware of any fact or condition that, individually or together with other facts or conditions, causes or constitutes a material breach of any of Seller's or Purchaser's representations and warranties as of the date of this Agreement, or if it becomes aware of the occurrence after the date of this Agreement of any fact or condition that, individually or together with other facts or conditions, would cause or constitute a material breach of any such representation or warranty had it been made as of the time of occurrence or discovery of such fact or condition. Should any such fact or condition require any change in the Seller Disclosure Schedule if the Seller Disclosure Schedule were dated the date of the occurrence or discovery of any such fact or condition, Seller will promptly deliver to Purchaser a supplement to the Seller Disclosure Schedule specifying such change. During the same period, each of Seller and Purchaser will promptly notify the other in writing of the occurrence of any material breach of any covenant of Seller or Purchaser or the occurrence of any event that may make the satisfaction of the conditions in Article VII impossible or unlikely. For the avoidance of doubt, no such disclosures or supplements shall affect Seller's or Purchaser's rights under Article VII or Article VIII. Furthermore, if (a) the conditions to Purchaser's obligation to close are not satisfied due to disclosures made by Seller in any supplement to the Seller Disclosure Schedule delivered by Seller to Purchaser as described above, (b) such disclosures reasonably should have been made in the Seller Disclosure Schedule delivered by Seller to Purchaser on the date hereof, and (c) as a result, the Closing does not occur, then Seller shall reimburse Purchaser's reasonable out-of-pocket costs (including attorneys' fees and bank commitment fees) incurred in connection with this Agreement and the transactions contemplated hereby, not to exceed \$1.5 million. If, however, Seller delivers to Purchaser a supplement to the Seller Disclosure Schedule as described above, and if the Closing occurs, the supplement shall be part and parcel of the Seller Disclosure Schedule for all purposes under this Agreement (including without limitation Articles II and IX), but Seller shall pay to Buyer one-half of the Damages that Purchaser demonstrates to Seller's reasonable satisfaction will be incurred as a result of such disclosure, not to exceed \$150,000. Notwithstanding anything herein to the contrary, but subject to Section 8.1(g) below, if Seller delivers any changes or additions to Schedule 2.13(a)(ii) or Schedule 2.16(b) within five Business Days after the execution and delivery hereof (the "Schedule Supplement Date"), such delivery shall for all purposes hereunder be deemed to have occurred on the date hereof.

4.11 No Negotiation. Until such time, if any, as this Agreement is terminated pursuant to Article VIII, Seller will not, and will cause each of Company and Subsidiaries not to,

directly or indirectly solicit, initiate or encourage any inquiries or proposals from, discuss or negotiate with, provide any non-public information to, or consider the merits of any unsolicited inquiries or proposals from, any party other than Purchaser or its Affiliates relating to any transaction involving the sale of the securities, business, assets or properties (other than in the ordinary course of the operation of the businesses or indirectly through any transaction involving Seller or any of its Affiliates that will not affect the sale of the Shares to Purchaser under this Agreement) of the Company or any Subsidiary, or any merger, consolidation, business combination or similar transaction involving the Company or any Subsidiary.

- 4.12 Resignations. At Closing, Seller shall cause the Company and each of the Subsidiaries to cause the persons listed on <u>Schedule 4.12</u> attached hereto to resign from their positions as officers, directors or other similar positions of the Company and the Subsidiaries.
- 4.13 Release. Seller shall execute and deliver to Purchaser at the Closing a release in substantially the form attached hereto as Exhibit A.
- 4.14 Employees. Prior to the Closing, Seller shall not, and shall cause each of the Company and the Subsidiaries and its Affiliates not to, discourage any employee of Seller listed on Schedule 6.1 attached hereto from transferring their employment to the Company or any Subsidiary if the Company or any Subsidiary seeks to employ such employee and such employee chooses to accept such employment or hinder any attempt by the Company or any Subsidiary to hire any such employee. For a period of 3 years following the Closing, neither Seller nor any of its Affiliates shall, directly or indirectly, solicit (other than by general solicitations not targeted at specific individuals, such as help-wanted advertising) any employee listed on Schedule 6.1 who accepts employment with the Company or any Subsidiary and remains in such employment. The parties acknowledge that Seller has a severance package for employees and will have job openings with Seller or one or more Affiliates of Seller and that neither (a) the existence of, informing the employees of the existence of, or making payments under said severance package, nor (b) the existence of, informing the employees of the existence of, or hiring the employees to fill said job openings will constitute a violation of this Section.

# 4.15 Noncompetition Agreement.

(a) For a period of 5 years after the Closing neither Seller nor any of its Affiliates shall, directly or indirectly, either individually or as a principal, partner, agent, employee, employer, consultant, stockholder, joint venturer, or investor, or as a director or officer of any corporation or association, or in any other manner or capacity whatsoever, construct a pipeline for the transportation of natural gas or convert a pipeline for some other resource into a natural gas pipeline, in either case located anywhere in the following counties located in the State of Missouri: Pike, Lincoln, St. Charles, Franklin, Washington, Crawford and Pulaski. The parties agree, however, that it shall not be a violation of this Section 4.15(a) if Seller or any of its Affiliates directly or indirectly constructs a pipeline for the transportation of natural gas to a merchant electricity generating facility owned or operated, in whole or in part, by Seller or any of its Affiliates.

- (b) The parties believe the covenants against competition contained in Section 4.15(a) are reasonable and fair in all respects, and are necessary to protect the interests of Purchaser. However, in case any one or more of the provisions or parts of a provision contained in Section 4.15(a) shall, for any reason, be held to be invalid, illegal or unenforceable in any respect in any jurisdiction, such invalidity, illegality or unenforceability shall not affect any other provision or part of a provision of Section 4.15(a), but Section 4.15(a) shall be reformed and construed in any such jurisdiction as if such invalid or illegal or unenforceable provision or part of a provision had never been contained herein and such provision or part shall be reformed so that it would be valid, legal and enforceable to the maximum extent permitted in such jurisdiction.
- Non-Disclosure. Neither Seller nor any of its Affiliates shall, at any time 4.16 within 3 years after the Closing, use or disclose, directly or indirectly, any Company Confidential Information (as defined below) to the detriment of Purchaser, the Company or any Subsidiary and without any reasonable commercial or regulatory justification; provided that, for the avoidance of doubt, this Section 4.16 shall not limit in any way the taking of any prudent action by Seller on behalf of its customers; provided, however, notwithstanding the foregoing phrase to the contrary, Company Confidential Information shall not be used in such action. As used herein, the term "Company Confidential Information" means non-public information that is disclosed to or known by Seller or an Affiliate of Seller as a consequence of its ownership and operation of the business and assets of the Company and its Subsidiaries, and (ii) relates to the business or assets of the Company or its Subsidiaries. Company Confidential Information shall not include (x) any information generally known in the industry on the date hereof (y) any information which becomes generally known in the industry after the date hereof through no fault of Seller or any of its Affiliates or (z) any information which is disclosed after the date hereof to Seller or an Affiliate not in violation of any rights of Purchaser known to Seller or such Affiliate. In the event that Seller or any of its Affiliates or anyone to whom such persons have transmitted the Company Confidential Information becomes legally compelled (by interrogatories, subpoena, civil investigative demand or similar process) to disclose any of the Company Confidential Information, Seller will provide to Purchaser prompt written notice of such demand so that Purchaser may seek a protective order or other appropriate remedy and/or waive compliance with the provisions of this Section. In the event that such protective order or other remedy is not obtained, or that Purchaser waives compliance with the provisions of this Agreement, Seller may and will furnish only that portion of the Company Confidential Information which is legally required and will exercise its best efforts to obtain reliable assurances that confidential treatment will be accorded the Company Confidential Information.
- 4.17 Data and Other Information. At the Closing, Seller shall deliver to Purchaser all data and information owned by the Company or a Subsidiary or used in the business of the Company and its Subsidiaries but in Seller's possession, including without limitation, customer files, books, records, contracts, maps, plats, surveys and photographs, whether in electronic format or otherwise; provided, however, that in the case of data and information used in the business of the Company and its Subsidiaries but in Seller's possession, such delivery shall not include data and information that (a) also relate to or are used in connection with other businesses of Seller or any of its other Affiliates, (b) cannot reasonably be redacted or otherwise edited to apply only to the business of the Company and its Subsidiaries,

and (c) are not necessary for the operation of the business of the Company and its Subsidiaries. Seller shall use reasonable commercial efforts to accommodate Purchaser's reasonable formatting requests in respect of such data and information in electronic format.

# ARTICLE V

#### TAX MATTERS

- 5.1 Definitions. For purposes of this Agreement, the following terms shall have the meanings ascribed to them below:
  - (a) "Adverse Consequences" means all actions, suits, proceedings, hearings, investigations, charges, complaints, claims, demands, injunctions, judgments, orders, decrees, rulings, damages, dues, penalties, fines, costs, amounts paid in settlement, obligations, taxes, Consolidated Income Taxes, Other Taxes, liens, losses, expenses, and fees, including court costs and attorney's or accountant's fees and expenses.
  - (b) "Consolidated Income Taxes" means Income Taxes filed on a consolidated, combined or unitary basis that include Seller or any of its subsidiaries (other than the Company or any of its Subsidiaries), on the one hand, and the Company or any of its Subsidiaries, on the other hand.
  - (c) "Income Taxes" means federal, state or local taxes measured by net income or capital gain.
  - (d) "Liability" means any liability (whether known or unknown, whether asserted or unasserted, whether absolute or contingent, whether accrued or unaccrued, whether liquidated or unliquidated, and whether due or to become due, including any liability for Consolidated Income Taxes, Other Taxes or Taxes.
  - (e) "Other Taxes" means Taxes other than Consolidated Income Taxes.
  - (f) "338 Forms" means all returns, documents, statements and other forms that are required to be submitted to any taxing authority in connection with the 338 Election (as hereinafter defined), including any "statement of Section 338 election" and Internal Revenue Service Form 8023 (together with any schedules or attachments thereto) that are required pursuant to any tax laws.
- 5.2 Section 338(h)(10) Election. Purchaser and Seller and Seller's Subsidiaries shall join in making a timely election under Section 338(h)(10) of the Code and the Treasury Regulations promulgated thereunder and any similar state statute or regulation with respect to the transactions contemplated hereunder involving the Company and its Subsidiaries (the "338 Election"). Purchaser shall execute any and all documents (including, without limitation, the 338 Forms) and take any and all action necessary to cause the 338 Election to become effective, and Seller shall use reasonable efforts to cooperate with Purchaser to prepare,

complete, and file such forms. In connection with the 338 Election, Purchaser and Seller shall cooperate to reach a mutually acceptable allocation of Purchase Price among the assets, which allocation shall be made in accordance with <u>Schedule 5.2</u> (which shall be consistent with Section 1.2 hereof) and shall be reasonable, based on fair market values and consistent with Section 338 of the Code and any comparable provisions of state or local income tax law or regulation. Unless otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code, Purchaser and Seller each shall report, act and file in all Income Tax respects and for all Income Tax purposes consistent with such allocation.

- 5.3 Tax Indemnification by Seller. Seller shall be liable for, and shall indemnify and hold harmless Purchaser, the Company, the Subsidiaries, and any successors thereto, and their respective officers, directors, employees and agents (collectively, the "Purchaser Indemnitees") from and against, all Adverse Consequences the Purchaser Indemnitees may incur or suffer resulting from, arising out of, or relating to:
  - (a) any Liability for Taxes of the Company or any of its Subsidiaries (not including any amounts for Taxes included in the Working Capital Adjustment in Section 1.5) (i) incurred with respect to any taxable period ending on or before the Closing Date, other than Income Taxes caused by the Liability of the Company and its Subsidiaries under Treasury Regulations Section 1.1502-6, (ii) accruing with respect to any taxable period resulting from a breach of any representations or warranties contained in Section 2.11 or Article V hereof, or (iii) incurred as a result of any actions taken pursuant to Section 1.2(a);
  - (b) the timely and proper making of the 338 Election, or Seller's inability or failure to timely or properly make the 338 Election;
  - (c) the settlement or other resolution (without the consent of Purchaser or the Company) of a proposed tax adjustment in respect of Taxes of the Company or any of its Subsidiaries which relates to a tax period ending on or before the Closing Date. For example, if Seller agrees in an income tax audit to reduce the depreciable basis of property acquired by the Company before the Closing Date, Seller shall be liable for any additional taxes due from the Company by reason of reduced depreciation deductions; and
  - (d) any and all sales taxes arising as a result of the sale and purchase of the Shares contemplated by this Agreement.
- 5.4 Tax Indemnification by Purchaser. Purchaser shall be liable for, and shall indemnify and hold harmless Seller, its Affiliates (other than the Company and its Subsidiaries), and any successors thereto, and their respective officers, directors, employees and agents (collectively, the "Seller Indemnitees") from and against, all Adverse Consequences the Seller Indemnitees may incur or suffer resulting from, arising out of, or relating to:
  - (a) any Liability for Taxes of Purchaser or any of its Affiliates (other than the Company or its Subsidiaries);

- (b) any Liability for Taxes of the Company or any of its Subsidiaries (other than such Taxes for which Seller is responsible under Section 5.3) accruing after the Closing Date; and
- (c) Purchaser's inability or failure to timely or properly make the 338 Election.
- 5.5 Indemnity Payments. With respect to any indemnity payment under this Sections 5.3 or 5.4, the parties agree to treat, to the extent permitted by law, all such payments as an adjustment to the consideration paid for the sale and transfer of the Shares of the Company hereunder, unless otherwise required pursuant to a "determination" within the meaning of Section 1313(a) of the Code. The parties also agree that the indemnifying party shall pay the indemnified party an additional or lesser amount sufficient to make such indemnified party whole, taking into account any deductions, credits or other tax benefits allowable with respect to the loss for which indemnification is required and any taxes payable on or with respect to such indemnity payment.
- 5.6 Survival. Any representation, warranty and agreement set out in Section 2.11 and/or Article V hereof shall survive until the later of the final resolution of any judicial or administrative proceeding involving the matter or the expiration of any applicable statute of limitations (including any suspensions, tollings or extensions thereof).
- 5.7 Defense of Tax Claim Subject to Seller's Indemnity. Whenever Purchaser, the Company or any Subsidiary receives any communication in which any taxing authority asserts a claim, makes an assessment, serves notice of a proposed audit or otherwise disputes the amount of Taxes in respect to which indemnity may be sought against Seller pursuant to Section 5.3 of this Agreement, Purchaser will promptly (and in any event within 20 days thereafter) notify Seller in writing, and Seller will have the right to control any resulting meetings, conferences or proceedings and to determine whether and when to settle any such claim, assessment or dispute to the extent such proceedings or determinations could affect the amount of Tax for which Seller is responsible. Purchaser and the Company shall reasonably cooperate in the preparation for and participation in any meetings, conferences or proceedings Seller controls. Seller agrees to pay to Purchaser and the Company all of their reasonable expenses which they shall incur in connection with such cooperation. Purchaser may request that Seller decline to take any further action with respect to any claim, demand or deficiency described herein, and Seller thereafter shall take no action, provided the Purchaser Indemnitees have notified Seller in writing that they waive their right to indemnification for any liability resulting therefrom. In the event Seller requests the Company to pay Taxes to a taxing authority prior to contesting the assessment thereof and thereafter to file a claim or suit for refund, Seller shall advance to the Company, on an interest-free basis, the amount of any such payment, after which the Company shall promptly pay the amount to the governmental authority as directed by Seller.
- 5.8 Defense of Tax Claim Subject to Purchaser's Indemnity. Whenever Seller receives any communication in which any taxing authority asserts a claim, makes an assessment, serves notice of a proposed audit or otherwise disputes the amount of Taxes in respect to which indemnity may be sought against Purchaser pursuant to Section 5.4 of this Agreement, Seller will

promptly (and in any event within 20 days thereafter) notify Purchaser in writing, and Purchaser and the Company will have the right to control any resulting meetings, conferences or proceedings and to determine whether and when to settle any such claim, assessment or dispute to the extent such proceedings or determinations could affect the amount of Tax for which Purchaser is responsible. Seller shall reasonably cooperate in the preparation for and participation in any meetings, conferences or proceedings Purchaser and the Company control. Purchaser agrees to pay to Seller all of its reasonable expenses which it shall incur in connection with such cooperation. Seller may request that the Purchaser and the Company decline to take any further action with respect to any claim, demand or deficiency described herein, and Purchaser and the Company thereafter shall take no action, provided the Seller Indemnitees have notified the Purchaser in writing that they waive their right to indemnification for any liability resulting therefrom. In the event Purchaser or the Company requests Seller to pay Taxes to a governmental authority prior to contesting the assessment thereof and thereafter to file a claim or suit for refund, Purchaser shall advance to Seller, on an interest-free basis, the amount of any such payment, after which the Seller shall promptly pay the amount to the governmental authority as directed by Purchaser.

# 5.9 Filing Responsibility.

- (a) Seller shall prepare (or cause to be prepared) and file (or cause to be filed) the consolidated federal income tax return of Seller's Affiliated group and all other consolidated, combined or unitary tax returns of Seller or its Tax Affiliates which include the Company, and shall report the operations of the Company in such Tax Returns for all taxable periods of the Company ending on or prior to the Closing Date. Except as otherwise provided in this Section 5.9, Purchaser shall be responsible for filing all Tax Returns required to be filed by or on behalf of the Company or any of its Subsidiaries after the Closing Date. Those Tax Returns which include any taxable period beginning before and ending after the Closing Date shall be filed on a basis which is consistent with the manner in which Seller or its Tax Affiliates filed such Tax Returns in the past, unless a contrary treatment is required by law. Purchaser shall not, and shall cause the Company and its Subsidiaries not to, file any amended returns for any tax period (or portion thereof) ending on or prior to the Closing Date, without Seller's prior written consent.
- (b) With respect to any Tax Return required to be filed by Purchaser for a taxable period of the Company or any Subsidiary which includes (but does not close on) the Closing Date, Purchaser shall provide Seller with copies of such completed Tax Return and a statement (the "Statement") certifying the amount, if any, of tax shown on such Tax Return that is allocable to Seller pursuant to Article V (subject to, for the avoidance of doubt, any reserves included in the Working Capital Adjustment) at least 20 Business Days prior to the due date for the filing of such Tax Return, and Seller and its authorized representatives shall have the right to review such Tax Return and Statement prior to the filing of such Tax Return. Seller and Purchaser agree to consult and resolve in good faith any issues arising as a result of the review of such Tax Return and Statement by Seller or its authorized representatives and to mutually consent to the filing of such Tax Return. No later than 3 days before the due date for payment of Taxes with respect

to such Tax Return, Seller shall pay to Purchaser an amount equal to the Taxes shown on the Statement as being allocable to Seller.

- 5.10 Tax Elections. No new elections with respect to Taxes or any changes in current elections with respect to Taxes affecting the Company shall be made by Seller after the date of this Agreement without the prior written consent of Purchaser.
- 5.11 Cooperation. Upon Purchaser's reasonable request, from time to time, Seller shall deliver or make available to Purchaser all information (including, without limitation, all work papers, schedules, memoranda and other information prepared by Seller or its Tax Affiliates, Subsidiaries and agents, relating to the assets of the Company) reasonably available to Seller and necessary to the preparation of the Company's Returns for periods ending after the Closing Date. Seller shall also provide to Purchaser, upon Purchaser's written request and after they become available to Seller, copies of the Company's separate pro forma federal income and state income tax returns for all tax years of the Company in which the Company was included in Seller's federal consolidated income tax return and for all tax periods ending on or prior to the Closing Date, together with any data or schedules reasonably necessary to support the computations and information shown on such returns. In the event of an audit of Purchaser or the Company by a taxing authority with respect to any Return for any taxable period or periods ending subsequent to the Closing Date, Seller shall provide Purchaser with such factual information which Seller possesses as Purchaser may reasonably request, in writing, with respect to the Company and shall otherwise provide such assistance as Purchaser may reasonably request in connection with such audit. Seller shall maintain and preserve its records with respect to the Company for at least six years from the Closing Date. Upon Seller's reasonable request, from time to time, Purchaser shall deliver or make available to Seller all information reasonably available to Purchaser and necessary to the preparation of Seller's Returns for periods ending on or before the Closing Date. In the event of an audit of the Company or any Subsidiary (or of the Seller in respect of the Company or any Subsidiary), Purchaser or the Company shall provide Seller with such factual information which Purchaser, the Company or any of its Subsidiaries possesses as Seller may reasonably request, in writing, with respect to the Company and its Subsidiaries and shall otherwise provide such assistance as Seller may reasonably request in connection with such audit.

# 5.12 Tax Refunds.

- (a) Seller shall be entitled to, and Purchaser agrees to promptly pay to Seller, all foreign, federal, state and local refunds and interest thereon (including, without limitation, as a credit or offset against any other Taxes) (collectively "Refunds"), if any, received by Purchaser or the Company to the extent attributable to (i) Taxes paid with respect to Returns of the Company for all taxable periods ending on or prior to the Closing Date, or (ii) any Taxes for which Seller has indemnified the Purchaser Indemnitees pursuant to this Agreement.
- (b) In the event and to the extent that Seller or any of its Tax Affiliates become entitled to a refund or credit of Taxes for any taxable period ending on or before the Closing Date attributable solely to the carryback of losses, credits or similar items

from any taxable period beginning after the Closing Date and attributable to the Company, Seller shall pay to Purchaser the amount of such refund or credit, within 30 days of receipt of such refund or credit together with interest accrued thereon at the rate and in the manner specified in Section 6621 of the Code for overpayments in respect to federal Income Taxes and at the rate provided by applicable law for overpayments in respect of Other Taxes, provided that (i) Purchaser shall indemnify and hold harmless Seller and its Tax Affiliates from and against any Liability, including interest and penalties, assessed against Seller or any of its Affiliates by reason of the reduction or disallowance and (ii) Seller shall not be obligated to use any such carryback item in preference to any carryback item which Seller or any of its Affiliates may otherwise use.

### 5.13 Income Tax Liabilities.

- (a) Subject to Section 5.2, the sale and transfer of the Shares of the Company hereunder shall be effective for state or local income tax purposes as of the Closing Date (including under the laws of any taxing jurisdiction which does not recognize a short tax period in respect of the sale and transfer of the Shares), and all income, deductions, losses, gains and credits accrued or incurred before the Closing Date shall be reportable by Seller.
- (b) Subject to Section 5.2, in order to apportion any state or local income taxes relating to a taxable period that includes, but that would not (except for this Section 5.13) end on the Closing Date, Seller and Purchaser will, to the extent permitted by applicable law, elect with the relevant state and local taxing authority to close the taxable period of the Company on the Closing Date. In any case where applicable law does not permit the Company to close its taxable year on the Closing Date, then such income taxes, if any, attributable to the taxable period of the Company that includes the Closing Date shall be allocated to Seller for the portion of the taxable period up to and including the Closing Date and to Purchaser for the portion of the taxable period subsequent to the Closing Date. For these purposes, such income taxes for the portion of the taxable period up to and including the Closing Date shall be determined in the case of a tax based on net income, on the basis of an interim closing of the books as of the Closing Date.
- (c) For the avoidance of doubt, except as otherwise provided herein, Seller and Purchaser shall be responsible for their own respective Income Taxes.
- 5.14 Article V to Control. To the extent that there is a conflict between any provision of Article V and any other provision of this Agreement, the provisions of Article V shall control.
- 5.15 Tax Sharing Agreements. All tax-sharing agreements or similar agreements with respect to or involving the Company shall be terminated prior to the Closing Date, and, after the Closing Date, the Company shall not be bound thereby or have any liability thereunder for amounts due in respect of periods prior to the Closing Date.

Plan of Liquidation. Notwithstanding anything to the contrary in this Agreement, Seller retains the right to adopt, prior to Closing, an agreement and plan of liquidation (the "Plan") with the Company in the form attached hereto as Schedule 5.16 to effect a complete liquidation of the Company for Federal Income Tax purposes under Section 332 of the Code. Purchaser acknowledges that as a result of the plan of liquidation so adopted, the Company's Subsidiary named Omega Pipeline Company, a Delaware corporation ("Omega"). will become a direct, wholly owned subsidiary of Seller, and, accordingly, prior to the Closing Date, Company will no longer own any shares of, or hold any other interest in, Omega or any of its assets or businesses. Purchaser also acknowledges that in the event Seller does not so adopt a plan of liquidation, Purchaser and the Company may take such other corporate actions as may be necessary to ensure that prior to the Closing Date the Company will no longer own any shares of, or hold any other interest in, Omega or any of its assets or businesses. In any event, for the avoidance of doubt, and notwithstanding any other provision of this Agreement to the contrary, the parties hereby affirm (a) their understanding and agreement that the Company will not, as of the Closing Date, own any shares of, or hold any other interest in, Omega or any of its assets or businesses, and (b) that for all purposes of this Agreement (other than this Section 5.16), Omega is not a Company Subsidiary. Seller shall be solely responsible for all Adverse Consequences as a result of the Plan.

### ARTICLE VI

#### **EMPLOYEE MATTERS**

- the Company and Purchaser shall take all steps necessary or appropriate so that each individual who is an employee of Seller seconded to the Company or a Company Subsidiary and who is listed on Schedule 6.1 of the Seller Disclosure Schedule (the "Company Employees") is offered a transfer to the employment of the Company or one of the Company Subsidiaries at substantially the same base cash compensation as such individuals were employed immediately before such transfer. Seller represents that no action by Seller will cause the Company Employees not to be "at-will" employees when employed by the Company. Seller shall be responsible for all of its employees who are not Company Employees. In the case of all Company Employees, such transfer shall be effective not later than the Closing, and the Company Employees' "Transfer Date" shall be the Closing Date. Purchaser shall retain all liabilities relating to its hiring decisions.
- Company and the Company Subsidiaries shall assume and be solely responsible for all wages, salary, overtime pay, bonuses, incentive pay, vacation pay and other cash compensation of Company Employees, regardless of when such compensation accrues or becomes payable and regardless of whether such compensation is for periods before, on or after the Closing Date to the extent that with respect to amounts prior to the Closing, such amounts are accrued as expenses in the calculation of Working Capital. The Company Employees will be ineligible for any incentive payment under Seller's Year 2001 Annual Incentive Plan ("AIP") should the Closing Date be prior to the pay-out date under the AIP. If the Company Employees are ineligible for an incentive payment under the AIP because the Closing Date occurs prior to the pay-out date,

Seller shall pay the Company Employees who accept employment with the Company or one of the Subsidiaries, within thirty (30) days following the Closing Date, an amount equal to the annual incentive award they would have earned based on a target level of personal and Seller performance under the AIP, pro-rated for the portion of the year 2001 the Company Employees worked for Seller or Seller's Affiliates, less any required withholdings. For the avoidance of doubt, no liability account relating to amounts payable under the AIP will be included in the calculation of Working Capital.

- 6.3 Employee Benefits Generally. Except as provided in Section 6.4 below, effective as of the Closing Date, the Company Employees shall cease to be active participants in the Plans. Purchaser, the Company and the Company Subsidiaries shall have no obligations and shall assume no liabilities with respect to the Employee Benefit Plans.
  - (a) From and after the Closing Date but except as provided in Section 6.4 below, the Company, the Company Subsidiaries and Purchaser shall be solely responsible for providing Company Employees with employee benefits (including without limitation welfare and savings benefits, including health, life, disability, and 401(k) benefits), consistent with benefits provided to other employees of Purchaser.
  - (b) For purposes of eligibility to participate and vesting under all compensation and benefit plans applicable to Company Employees after the Closing, Purchaser, the Company and the Company Subsidiaries shall give Company Employees credit for all service with Seller and its Affiliates before the Closing Date.
- 6.4 Welfare Benefits and Workers Compensation. Without limiting the generality of the above provisions, this Section 6.4 contains certain specific provisions regarding the provision of benefits under "welfare plans" as defined in Section 3(l) of ERISA ("Welfare Benefit Plans") and of workers compensation benefits.
  - (a) Effective as of 12:01 a.m. on the first day of the first calendar month following the Closing Date (the "Plan Date"), Purchaser shall cause the Company Employees to be covered by Welfare Benefit Plans sponsored by Purchaser and/or its Affiliates (and Seller shall continue to provide coverage for Company Employees prior to such Date subject to the terms of such plans).
  - (b) Seller shall be responsible for (i) claims of Company Employees and their eligible beneficiaries and dependents for workers compensation and under Welfare Benefit Plans that are incurred on or before the Plan Date, and (ii) claims relating to continuation of health coverage required pursuant to Section 4980B of the Code or Part 6 of Subtitle B of Title I of ERISA ("COBRA Coverage") attributable to "qualifying events" occurring on or before the Plan Date with respect to any Company Employees and their eligible beneficiaries and dependents. Purchaser, the Company and the Company Subsidiaries shall be solely responsible for (A) claims of Company Employees and their eligible beneficiaries and dependents for workers compensation benefits and claims under Welfare Benefit Plans that are incurred after the Plan Date, and (B) claims relating to COBRA Coverage or any applicable continuation provisions of state law

attributable to "qualifying events" occurring after the Plan Date with respect to Company Employees and their beneficiaries and dependents. For purposes of the foregoing, a medical/dental claim shall be considered incurred on the date when the medical services are rendered or medical supplies are provided, and not when the condition arose or when the course of treatment began; provided, that claims relating to a hospital confinement that begins before the Plan Date but continues thereafter shall be treated as incurred before the Plan Date. A disability or workers compensation claim shall be considered incurred before the Plan Date if the illness or injury giving rise to the claim occurs before the Plan Date.

(c) Each Company Employee shall be immediately eligible to participate, without any waiting time, in any and all Welfare Benefit Plans sponsored by Purchaser and its Affiliates for the benefit of Company Employees (such plans, collectively, the "New Welfare Plans") to the extent coverage under such New Plan replaces coverage under a comparable Welfare Benefit Plan sponsored by Seller in which such Company Employee was previously participating (such plans, collectively, the "Old Welfare Plans"). For purposes of each New Welfare Plan providing medical, dental, pharmaceutical and/or vision benefits, Purchaser shall use reasonable commercial efforts to cause all pre-existing condition exclusions and actively-at-work requirements of such New Welfare Plan to be waived for Company Employees and their eligible beneficiaries and dependents, and Purchaser shall cause any eligible expenses incurred by any Company Employee and his or her eligible beneficiaries and dependents during the portion of the plan year of the Old Welfare Plan ending on the date such Company Employee's participation in the corresponding New Welfare Plan begins to be taken into account under such New Welfare Plan for purposes of satisfying all deductible, coinsurance and maximum out-of-pocket requirements applicable to such employee and his or her eligible beneficiaries and dependents for the applicable plan year as if such amounts had been paid in accordance with such New Welfare Plan.

#### ARTICLE VII

#### CONDITIONS TO OBLIGATIONS TO CLOSE

7.1 Conditions to Purchaser's Obligation to Close. Purchaser's obligation to consummate the purchase of the Shares on the terms specified herein shall be subject to the satisfaction or waiver on or prior to the Closing Date of all of the following conditions:

### (a) Representations, Warranties and Covenants of Seller.

(i) The representations and warranties of Seller contained in this Agreement shall be true and correct in all respects on and as of the Closing Date with the same effect as though such representations and warranties had been made on and as of such date (except for representations and warranties expressly made as of an earlier date, in which case as of such date or time), except for such failures to be true and correct which, without regard to any materiality qualifications in such representations and warranties, would not, individually or in

the aggregate, result in a material adverse effect on the business or operations of the Company and the Subsidiaries, taken as a whole. For purposes of this Section 7.1(a), the term "material" shall mean in relation to a matter which is quantifiable in monetary terms, an impact which can reasonably be expected to be in an amount greater than \$300,000 individually or in the aggregate.

- (ii) The covenants and Agreements of Seller to be performed on or before the Closing Date in accordance with this Agreement shall have been duly performed in all material respects. In addition and notwithstanding the foregoing, the covenants contained in the Notes to <u>Schedule 2.13B</u> must be performed in full prior to Closing.
- (iii) Purchaser shall have received at the Closing a certificate dated the Closing Date and validly executed on behalf of Seller by an officer of Seller to the effect that the conditions specified in clauses (i) and (ii) of this Section 6.1(a) have been satisfied.
- (b) <u>HSR Waiting Period</u>. Any waiting periods applicable to the transactions contemplated by this Agreement under applicable United States antitrust or trade regulation laws and regulations, including, without limitation, under the HSR Act, shall have expired or been terminated.
- (c) <u>No Injunction</u>. At the Closing Date, (i) there shall be no statute, ordinance, rule, regulation, injunction, writ, judgment, judicial decision, order or decree of any nature of any Governmental Entity that is in effect that restrains, prohibits, prevents, delays, makes illegal or otherwise interferes with the consummation of the transactions contemplated by this Agreement; and (ii) no action or proceeding before any Governmental Entity shall have been instituted to restrain or prohibit Purchaser's acquisition of the Shares or that challenges, or seeks damages or other relief in connection with, the transactions contemplated herein.
- (d) <u>Governmental Entity Approval</u>. Any required consent, waiver, approval or authorization by any Governmental Entity of the transactions contemplated by this Agreement shall have been obtained.
- (e) <u>Counsel Opinion</u>. Seller's counsel shall have delivered an opinion in substantially the form attached hereto as Exhibit B to Purchaser and Purchaser's counsel, addressed to Purchaser and Purchaser's lender.
- (f) <u>Transition Services Agreement</u>. The Company and Seller shall have executed and delivered a Transition Services Agreement in substantially the form attached hereto as Exhibit C.
- 7.2 Conditions to Seller's Obligation to Close. Seller's obligation to consummate the transactions contemplated by this Agreement shall be subject to the satisfaction on or prior to the Closing Date of all of the following conditions:

- (a) <u>Representations, Warranties and Covenants of the Company and</u> Purchaser.
  - (i) The representations and warranties of the Company and Purchaser contained in this Agreement shall be true and correct in all material respects on and as of the Closing Date with the same effect as though such representations and warranties had been made on and as of such date (except for representations and warranties expressly made as of an earlier date, in which case as of such date or time).
  - (ii) The covenants and agreements of the Company and Purchaser to be performed on or before the Closing Date in accordance with this Agreement shall have been duly performed in all material respects.
  - (iii) Seller shall have received at the Closing a certificate dated the Closing Date and validly executed on behalf of Purchaser by an officer of Purchaser to the effect that the conditions specified in clauses (i) and (ii) of this Section 6.2(a) have been satisfied.
- (b) <u>HSR Waiting Periods</u>. Any waiting periods applicable to the transactions contemplated by this Agreement under applicable United States antitrust or trade regulation laws and regulations, including, without limitation, under the HSR Act, shall have expired or been terminated.
- (c) No Injunction. At the Closing Date, (i) there shall be no statute, ordinance, rule, regulation, injunction, writ, judgment, judicial decision, order or decree of any nature of any Governmental Entity that is in effect that restrains, prohibits, prevents, delays, makes illegal or otherwise interferes with the consummation of the transactions contemplated by this Agreement; and (ii) no action or proceeding before any Governmental Entity shall have been instituted to restrain or prohibit Purchaser's acquisition of the Shares or that challenges, or seeks damages or other relief in connection with, the transactions contemplated herein.
- (d) <u>Governmental Entity Approval</u>. Any required consent, waiver, approval or authorization by any Governmental Entity of the transactions contemplated by this Agreement shall have been obtained, and such approval shall not require or be conditioned upon any requirement that Seller or any of its Affiliates provide any undertaking or agreement, or take or refrain from taking any other action, which, in Seller's reasonable judgment, would, individually or in the aggregate, adversely affect the continuing business of Seller or any of its Affiliates.

#### ARTICLE VIII

#### **TERMINATION**

- 8.1 Termination. This Agreement may be terminated at any time prior to the Closing by:
  - (a) The mutual written consent of Seller and Purchaser;
  - (b) Either Seller or Purchaser if the Closing has not occurred by the close of business on December 31, 2001, provided the failure to consummate the transactions contemplated by this Agreement did not result from the willful failure by the party seeking termination of this Agreement to fulfill any undertaking or commitment provided for herein that is required to be fulfilled prior to the Closing;
  - (c) Either Seller or Purchaser if the other shall have breached or failed to perform any of its respective representations, warranties, covenants or other agreements contained in this Agreement, which breach or failure to perform (i) would give rise to the failure of a condition set forth in Section 7.1(a) or 7.2(a), as applicable, and (ii) cannot be or has not been cured within 30 days after the giving of written notice to Purchaser or Seller, as applicable (and, specifically, the parties acknowledge that if Purchaser cannot obtain financing for any reason other than a material breach hereof by Seller, and if, as a result, Purchaser cannot pay the Purchase Price at Closing, the failure to pay the Purchase Price will constitute a breach of this Agreement by Purchaser, and provided, that if Purchaser fails to deliver the Deposit, this Agreement shall terminate if Seller has not received the Deposit on or before 5:00 p.m. on the 12th Business Day after the execution and delivery hereof by all parties without liability of any type on the part of Purchaser);
  - (d) Either Seller or Purchaser in the event that any Law of any Governmental Entity becomes effective (and final and nonappealable) permanently restraining, enjoining or otherwise prohibiting or making illegal or otherwise prohibiting the consummation of the transactions contemplated by this Agreement, upon notification of the non-terminating party by the terminating party;
  - (e) By Seller within 10 Business Days after the execution and delivery hereof if Seller, in its sole discretion, determines the approval of the transaction contemplated hereby by any utility regulatory body is either necessary or desirable;
  - (f) By Purchaser within 10 Business Days after the execution and delivery hereof if Purchaser, in its sole discretion, determines the relationship after the Closing with Laclede Gas Company will likely not be acceptable to Purchaser; or
  - (g) By Purchaser within 10 Business Days after the execution and delivery hereof if Purchaser, in its sole discretion, is not satisfied with either <u>Schedule</u> 2.13 (a)(ii) or <u>Schedule</u> 2.16(b).

8.2 Procedure and Effect of Termination. In the event of termination of this Agreement by either or both of Seller and Purchaser pursuant to Section 8.1, written notice thereof shall forthwith be given by the terminating party to the other party hereto, and this Agreement shall thereupon terminate and become void and have no effect, and the transactions contemplated by this Agreement shall be abandoned without further action by the parties hereto, except that the provisions of Section 4.1(a) and this Section 8.2 shall survive the termination of this Agreement; provided, however, that such termination shall not relieve any party hereto of any liability for any willful breach of this Agreement, except as otherwise provided herein. In the event Purchaser terminates this Agreement pursuant to Section 8.1(c) as a result of a breach or failure to perform by Seller, the Deposit (and all interest earned thereon) will be returned to Purchaser. In the event Seller terminates this Agreement pursuant to Section 8.1(c) as a result of a breach or failure to perform by Purchaser, Section 1.3 shall govern. If Seller terminates this Agreement pursuant to Section 8.1(b) as a result of the failure of the condition to Seller's obligation to consummate the transactions contemplated by this Agreement set forth in Section 7.2(d) to be satisfied, and if such failure is due to the imposition by a Governmental Entity in connection with any required consent, waiver, approval or authorization of a requirement or condition that does not either (a) require Seller or any of its Affiliates to dispose of any assets or modify or cease any business or businesses or operating practice (including without limitation any limitation on prices or rates) or (b) otherwise result in an adverse effect, cost or damage on Seller and its Affiliates reasonably determined by Seller to be equal to or greater than \$1 million in the aggregate, or for which monetary damages cannot reasonably be measured, then Seller shall reimburse Purchaser's reasonable out-of-pocket costs (including attorneys' fees and bank commitment fees) incurred in connection with this Agreement and the transactions contemplated hereby, not to exceed \$1.5 million and return the Deposit (and all interest earned thereon). Notwithstanding the preceding sentence or anything else in this Agreement to the contrary, in the event that the condition to Seller's obligation to consummate the transactions contemplated by this Agreement set forth in Section 7.2(d) is not satisfied due to the imposition by a Governmental Entity in connection with any required consent, waiver, approval or authorization of a requirement or condition that results in an adverse effect, cost or damage on Seller and its Affiliates reasonably determined by Seller to be less than \$1 million, such condition shall be deemed satisfied for all purposes under this Agreement if Purchaser agrees to pay Seller, by wire transfer of immediately available funds together with the Purchase Price, the amount of such adverse effect, cost or damage. If this Agreement is terminated as provided herein, all filings, applications and other submissions made pursuant hereto shall, to the extent practicable, be withdrawn from the agency or other persons to which they were made by the party making such filing, application or other submission.

#### ARTICLE IX

### INDEMNIFICATION; SURVIVAL

9.1 Indemnification by Purchaser. Subject to the provisions of this Article IX, Purchaser shall indemnify and hold harmless Seller and its Affiliates and their respective directors, officers, and employees from and against any and all Damages (as hereinafter defined) incurred by Seller and its Affiliates in connection with (a) a breach of any representation or warranty made by Purchaser hereunder or in any schedule, exhibit or other document attached to

or delivered pursuant to this Agreement, (b) any failure by Purchaser (or, following the Closing, the Company) to perform any covenant or other agreement hereunder, and (c) any action, suit, proceeding or claim incident to any of the foregoing. As used in this Agreement, "Damages" shall mean claims, liabilities, damages (including incidental and consequential damages), penalties, judgments, assessments, losses, costs and expenses, including reasonable attorneys' fees and expenses, incurred by the party seeking indemnification under this Agreement, net of (i) any insurance proceeds which such party actually receives in respect of such matter and (ii) any indemnity payments which such party receives from parties other than the party against whom such claim is asserted under this Agreement.

- 9.2 Indemnification by Seller. Subject to the provisions of this Article IX, Seller shall indemnify and hold harmless Purchaser and its Affiliates and their respective directors, officers and employees from and against any and all Damages incurred by Purchaser or its Affiliates in connection with (a) a breach of any representation or warranty made by Seller or the Company hereunder or in any schedule, exhibit or other document attached to or delivered pursuant to this Agreement, (b) any failure by Seller (or, prior to the Closing, the Company) to perform any covenant or agreement hereunder, (c) all Damages and all fines, penalties, interest and refunds which may be imposed by or required by any governmental or regulatory body or court as a result of any amounts charged by the Company prior to Closing to its customers being deemed by such governmental or regulatory body to be in excess of amounts which could legally be charged by the Company, (d) the Grynberg and Quinque lawsuits set forth on Schedule 2.10 hereto, (e) breach of the covenants set forth in the Notes to Schedule 2.13B, (f) the Environmental claims disclosed in Schedule 2.16(b) and (g) any action, suit, proceeding or claim incident to any of the foregoing.
- 9.3 Indemnification Procedure. The party or parties making a claim for indemnification under this Article IX shall be, for the purposes of this Agreement, referred to as the "Indemnified Party" and the party or parties against whom such claims are asserted under this Article IX shall be for the purposes of this Agreement, referred to as the "Indemnifying Party". From and after the Closing, the Company and the Subsidiaries shall be deemed an Affiliate of Purchaser for the purposes of this Article IX. All claims by any Indemnified Party under this Article IX shall be asserted and resolved as follows:
  - (a) In the event that (i) any claim, demand or proceeding is asserted or instituted by any person other than the parties to this Agreement or their Affiliates which could give rise to Damages for which an Indemnifying Party could be liable to an Indemnified Party under this Agreement (such claim, demand or proceeding, a "Third Party Claim") or (ii) any Indemnified Party under this Agreement shall have a claim to be indemnified by any Indemnifying Party under this Agreement which does not involve a Third Party Claim (such claim, a "Direct Claim"), the Indemnified Party shall with reasonable promptness send to the Indemnifying Party a written notice specifying the nature of such claim, together with information reasonably available to the Indemnified Party with respect to such claim (a "Claim Notice"), provided that a delay in notifying the Indemnifying Party shall not relieve the Indemnifying Party of its obligations under this Agreement except to the extent that (and only to the extent that) such failure shall have caused actual material prejudice to the Indemnifying Party.

- In the event of a Third Party Claim, the Indemnifying Party shall be entitled to appoint counsel of the Indemnifying Party's choice (who must be reasonably acceptable to the Indemnified Party) at the expense of the Indemnifying Party to represent the Indemnified Party and any others the Indemnifying Party may reasonably designate in connection with such claim, demand or proceeding (in which case the Indemnifying Party shall not thereafter be responsible for the fees and expenses of any separate counsel retained by any Indemnified Party except as set forth below). Notwithstanding an Indemnifying Party's election to appoint counsel to represent an Indemnified Party in connection with a Third Party Claim, an Indemnified Party shall have the right to employ separate counsel, and the Indemnifying Party shall bear the reasonable fees, costs and expenses of such separate counsel if, and only if, (i) the use of the counsel selected by the Indemnifying Party to represent the Indemnified Party would present such counsel with a conflict of interest or (ii) the Indemnifying Party shall not have employed counsel to represent the Indemnified Party within thirty (30) days after notice of the institution of such Third Party Claim. If requested by the Indemnifying Party, the Indemnified Party agrees to cooperate, at the Indemnifying Party's expense, with the Indemnifying Party and its counsel in contesting any claim, demand or proceeding, or, if appropriate and related to the claim, demand or proceeding in question, in making any counterclaim against the person asserting the Third Party Claim, or any cross-complaint against any person. No Third Party Claim may be settled or compromised by the Indemnified Party without the prior written consent of the Indemnifying Party, which consent shall not be unreasonably withheld or delayed. In the event any Indemnified Party settles or compromises or consents to the entry of any judgment with respect to any Third Party Claim without seeking to obtain the prior written consent of the Indemnifying Party, each Indemnified Party shall be deemed to have waived all rights against the Indemnifying Party for indemnification under this Article IX. No Third Party Claim may be settled or compromised by the Indemnifying Party without the prior written consent of the Indemnified Party, which consent shall not be unreasonably withheld or delayed, unless the Indemnified Party receives a complete release and is not otherwise adversely (either currently or prospectively) affected by such release.
- (c) In the event of a Direct Claim, the Indemnifying Party shall notify the Indemnified Party within 30 Business Days of receipt of a Claim Notice whether or not the Indemnifying Party disputes such claim. If the Indemnifying Party does not provide written notice providing in reasonable detail its reason for disputing a Direct Claim in such 30-day period, then the Indemnifying Party shall be deemed to accept the validity of the Direct Claim and all amounts claimed therein by the Indemnified Party and shall pay such amounts to the Indemnified Party promptly.
- (d) From and after the delivery of a Claim Notice under this Agreement, at the reasonable request of the Indemnifying Party, each Indemnified Party shall grant the Indemnifying Party and its representatives all reasonable access to the books, records and properties of such Indemnified Party and upon reasonable notice to the extent reasonably related to the matters to which the Claim Notice relates. All such access shall be granted during normal business hours and shall be granted under conditions which will not unreasonably interfere with the business and operations of such

- Indemnified Party. The Indemnifying Party will not, and shall require that its representatives do not, use (except in connection with such Claim Notice) or disclose to any third party other than the Indemnifying Party's representatives (except as may be required by applicable Law of any Governmental Entity) any information obtained pursuant to this Section 9.3.
- 9.4 Survival. Subject to Section 5.6, the representations and warranties of the parties contained in this Agreement shall survive the Closing and remain enforceable until 18 months after the Closing (e.g., if the Closing occurs on June 30, 2001, then the representations and warranties will expire on December 31, 2002); except that the representations and warranties of the parties contained in Sections 2.1, 2.2, 2.3, 2.25, 3.1, 3.2 and 3.3 shall not be subject to a contractual limitation period; the representations and warranties contained in Sections 2.13 and 2.16 shall expire on the fifth anniversary of the Closing Date. No claim for indemnity under this Article IX for any breach of a representation or warranty may be brought unless the appropriate Claim Notice shall have been delivered to the Indemnifying Party prior to the expiration of the survival period.
- Purchaser's Indemnification Limitation. Except as otherwise provided 9.5 herein, Seller's obligation to indemnify Purchaser for breaches of representations or warranties by Seller (or any action, suit, proceeding or claim incident to such breaches) as provided in Section 9.2 shall not become effective until the aggregate of all Damages sustained by Purchaser as described in Section 9.2 shall have exceeded \$50,000 (the "Basket") (Damages pursuant to Section 9.2(d) and 9.2(e) are not included in the Basket and are indemnified from the first dollar). If the total amount of Damages sustained by Purchaser as described in Section 9.2 exceeds the Basket then Purchaser shall be entitled to assert claims under this Article IX for indemnification for breaches of representations or warranties by Seller (or any action, suit, proceeding or claim incident to such breaches) for the amount of such Damages in excess of the Basket; provided that Seller's obligation under this Article IX shall not exceed \$5 million in the aggregate except for claims under Sections 2.1, 2.2, 2.3, 2.10, 2.11, 2.12, 2.13, 2.25, 2.26, 9.2(c), 9.2(d), 9.2(e) and Article V for which Purchaser shall receive first dollar recovery for all Damages up to the Final Adjusted Purchase Price. Indemnification pursuant to this Article IX shall be Purchaser's exclusive remedy for any Damages arising after the Closing under this Agreement either in contract, tort, or otherwise.
- 9.6 Seller's Indemnification Limitation. Except as otherwise provided herein, Purchaser's obligation to indemnify Seller for breaches of representation or warranties by Purchaser (or any action, suit, proceeding or claim incident to such breaches) as provided in Section 9.1 shall not become effective until the aggregate of all Damages sustained by Seller as described in Section 9.1 shall have exceeded \$50,000. If the total amount of Damages sustained by Seller as described in Section 9.1 exceeds \$50,000, then Seller shall be entitled to assert claims under this Article IX for indemnification for breaches of representations or warranties by Purchaser (or any action, suit, proceeding or claim incident to such breaches) for the amount of such excess; provided that Purchaser's obligation under this Article IX shall not exceed \$5 million in the aggregate except for claims under Sections 3.1, 3.2 and 3.3 for which Seller shall receive first dollar recovery for all Damages up to the Final Adjusted Purchase Price.

Indemnification pursuant to this Article IX shall be Seller's exclusive remedy for any Damages arising after the Closing under this Agreement either in contract, tort, or otherwise.

#### ARTICLE X

#### **MISCELLANEOUS**

- 10.1 Counterparts. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties.
- 10.2 Governing Law; Jurisdiction and Forum; Waiver of Jury Trial. This Agreement shall be governed by and construed in accordance with the laws of the State of Missouri without reference to the choice of law principles thereof.
  - (a) Each of Seller and Purchaser irrevocably submits to the jurisdiction the federal court sitting in the Western District of Missouri in any action arising out of or relating to this Agreement, and hereby irrevocably agrees that all claims in respect of such action may be heard and determined in such Missouri state or federal court. Each of Seller and Purchaser hereby irrevocably waives, to the fullest extent it may effectively do so, the defense of an inconvenient forum to the maintenance of such action or proceeding. The parties further agree, to the extent permitted by law, that final and unappealable judgment against any of them in any action or proceeding contemplated above shall be conclusive and may be enforced in any other jurisdiction within or outside the United States by suit on the judgment, a certified copy of which shall be conclusive evidence of the fact and amount of such judgment.
  - (b) To the extent that Purchaser or Seller has or hereafter may, acquire any immunity from jurisdiction of any court or from any legal process (whether through service or notice, attachment prior to judgment, attachment in aid of execution, execution or otherwise) with respect to itself or its property, each of Purchaser and Seller hereby irrevocably waives such immunity in respect of its obligations with respect to this Agreement.
  - (c) Each party waives, to the fullest extent permitted by applicable law, any right it may have to a trial by jury in respect of any action, suit or proceeding arising out of or relating to this Agreement. Each party certifies that it has been induced to enter into this Agreement or instrument by, among other things, the mutual waivers and certifications set forth above in this Section 10.2.
- 10.3 Entire Agreement. This Agreement (including agreements incorporated herein and the Confidentiality Agreement) and the schedules and exhibits hereto contain the entire agreement between the parties with respect to the subject matter hereof and there are no agreements, understandings, representations or warranties between the parties other than those set forth or referred to herein. Except for Sections 9.1 and 9.2, which are intended to benefit, and

to be enforceable by, any of the Indemnified Parties, this Agreement is not intended to confer upon any person not a party hereto (and their successors and assigns) any rights or remedies hereunder.

- 10.4 Expenses. Except as set forth in this Agreement, whether or not the transactions contemplated this Agreement are consummated, all legal and other costs and expenses incurred in connection with this Agreement and the transactions contemplated by this Agreement shall be paid by the party incurring such costs and expenses.
- 10.5 Notices. All notices and other communications to be given to any party hereunder shall be sufficiently given for all purposes hereunder if in writing and delivered by hand, courier or overnight delivery service or three days after being mailed by certified or registered mail, return receipt requested, with appropriate postage prepaid, or when received in the form of a telegram or facsimile, receipt confirmed, and shall be directed to the address or facsimile number set forth below (or at such other address or facsimile number as such party shall designate by like notice):

#### If to Seller:

UtiliCorp United Inc. 20 West Ninth St. Kansas City, MO 64105 Attention: General Counsel Fax No.: (816) 783-5175

#### If to Purchaser:

Gateway Pipeline Company, Inc. 7662 Davis Peak Road
Littleton, CO 80127
Attention: David Ries
Fax No.: (303) 933-6895

# With a copy to:

Bryan Cave LLP 3500 One Kansas City Place 1200 Main Street Kansas City, Missouri 64105 Attention: James P. Pryde, Esq. Fax No.: (816) 374-3300

10.6 Successors and Assigns. This Agreement shall be binding upon and inure to the benefit of the parties hereto and their respective successors and permitted assigns; provided, however, that no party hereto will assign its rights or delegate any or all of its obligations under this Agreement without the express prior written consent of each other party

hereto; provided, further, however, that Purchaser may assign its rights hereunder to any lender to Purchaser.

- 10.7 Headings; Definitions. The section and article headings contained in this Agreement are inserted for convenience of reference only and will not affect the meaning or interpretation of this Agreement. All references to Sections or Articles contained herein mean Sections or Articles of this Agreement unless otherwise stated. All capitalized terms defined herein are equally applicable to both the singular and plural forms of such terms.
- amended except by an instrument or instruments in writing signed by the party against whom enforcement of any such modification or amendment is sought. Either party hereto may, only by an instrument in writing, waive compliance by the other party hereto with any term or provision of this Agreement on the part of such other party hereto to be performed or complied with. The waiver by any party hereto of a breach of any term or provision of this Agreement shall not be construed as a waiver of any subsequent breach. Except as otherwise expressly provided herein, no failure to exercise, delay in exercising or single or partial exercise of any right, power or remedy by any party, and no course of dealing between the parties, shall constitute a waiver of any such right, power or remedy.
- 10.9 Schedules. The disclosure or inclusion of any matter or item on any Schedule to the Seller Disclosure Schedule shall not be deemed an acknowledgment or admission that any such matter or item is required to be disclosed. Seller and the Company shall not be prejudiced in any manner whatsoever by, and no presumptions shall be created by virtue of, any disclosure of any matter in the Seller Disclosure Schedule which is not expressly required to be disclosed under this Agreement.
- 10.10 Severability. If any provision of this Agreement shall be held invalid, illegal or unenforceable, the validity, legality or enforceability of the other provisions of this Agreement shall not be affected thereby, and there shall be deemed substituted for the provision at issue a valid, legal and enforceable provision as similar as possible to the provision at issue.
- 10.11 Interpretation. For the purposes of this Agreement, "to Seller's knowledge" shall mean the actual knowledge of the persons identified on Schedule 10.11 of the Seller Disclosure Schedule after inquiry. The phrase "including" shall be deemed to be followed by "without limitation." In the event an ambiguity or question of intent or interpretation arises, this Agreement shall be construed as if drafted jointly by the parties and no presumption or burden of proof shall arise favoring or disfavoring any party by virtue of the authorship of any provisions of this Agreement. For purposes of this Agreement, the word "person" shall be broadly construed to include individuals, corporations, partnerships (of whatever kind), limited liability companies, trusts, associations, any other types of entities, or any group or combination of any of the foregoing.
- 10.12 Specific Performance. The parties hereto agree that irreparable damage would occur in the event that any party fails to consummate the transactions contemplated by this Agreement in accordance with the terms of this Agreement and that the parties shall be entitled

to specific performance in such event, in addition to any other remedy or law or in equity. The parties hereto also agree that Purchaser's rights under Sections 4.15 and 4.16 are special and unique and that any violation thereof would not be adequately compensated by money damages, and Seller hereby grants Purchaser the right to specifically enforce (including injunctive relief where appropriate) the terms of Sections 4.15 and 4.16.

IN WITNESS WHEREOF, this Agreement has been signed by or on behalf of each of the parties by their duly authorized representatives as of the date first above written.

GATEWAY PIPELINE COMPANY, INC.

By: 下

Title: Presiden

UTILICORP PIPELINE SYSTEMS, INC.

By:<u>△</u>≥

Title: `

UTILICORP UNITED INC.

By:

Title: Sr. VICE-President

# Exhibit List

- Release A.
- B.
- Seller's opinion of counsel Transition Service Agreement C.

. -×. • . .

#### RELEASE

THIS RELEASE is entered into as of [•], 2001, by UtiliCorp United Inc., a Delaware corporation ("UCU") in favor of UtiliCorp Pipeline Systems, Inc., a Delaware corporation ("UPL"), Missouri Pipeline Company, a Delaware corporation ("MPC"), and Missouri Gas Company, a Delaware corporation ("MGC").

WHEREAS, MPC and MGC are wholly owned subsidiaries of UPL; and

WHEREAS, UPL is a wholly owned subsidiary of UCU; and

WHEREAS, UCU has agreed to sell all of the issued and outstanding shares of capital stock of UPL to Gateway Pipeline Company, Inc., a Delaware corporation ("Gateway"), pursuant to that certain Stock Purchase Agreement, dated February [•], 2001, by and among UCU, Gateway, and UPL (the "Stock Purchase Agreement"); and

WHEREAS, it is a condition to Gateway's obligation to close the transactions contemplated by the Stock Purchase Agreement that UtiliCorp execute and deliver this Release to Gateway;

NOW, THEREFORE, for good and valuable consideration, the receipt and sufficiency of which are hereby acknowledged, UtiliCorp hereby agrees as follows:

- 1. Release. From and after the date of the closing of the transactions contemplated by the Stock Purchase Agreement (the "Closing Date"), UCU, on behalf of itself and all entities controlled by UCU, hereby releases and forever discharges UPL (including TransMississippi Pipeline Company, which merged into UPL on June 30, 1999), MPC, MGC, and each of their directors, officers, shareholders, and employees from all liabilities, actions, causes of action, claims, and demands whatsoever, whether based upon statute or common law, and whether known or unknown, arising prior to the Closing Date.
- 2. Exclusions. For the avoidance of doubt, the release granted in Section 1 above shall not apply to obligations accruing (a) under the Stock Purchase Agreement, (b) under the Transition Services Agreement entered into on the date hereof by UCU and Gateway in connection with the closing of the transactions contemplated by the Stock Purchase Agreement, (c) on and after the Closing Date under the contracts that were entered into before the Closing Date by UPL, MPC, or MGC on the one hand and UCU or any entity controlled by UCU on the other hand and that are listed in the Seller's Disclosure Schedule (as defined in the Stock Purchase Agreement).

3. Governing Law. This Release shall be governed by and construed in accordance with the laws of the State of Missouri without reference to the choice of law principles thereof.

IN WITNESS WHEREOF, this Release has been signed on behalf of UCU by its duly authorized representative as of the date first above written.

UtiliCorp United Inc.

Name:		 _
Title:		
	 -	 _

### [UtiliCorp Letterhead]

[•], 2001

Gateway Pipeline Company, Inc. 7662 Davis Peak Road Littleton, CO 80127 Attention: David Ries

### [Address of Gateway's Lender]

Re: UtiliCorp United Inc. and UtiliCorp Pipeline Systems, Inc.

Dear Mr. Ries

The Office of the General Counsel of UtiliCorp United Inc., a Delaware corporation ("<u>UtiliCorp</u>"), has acted as counsel for UtiliCorp and UtiliCorp Pipeline Systems, Inc., a Delaware corporation ("<u>UPL</u>"), in connection with the Stock Purchase Agreement dated as of February [•], 2001 (the "<u>Agreement</u>") among UtiliCorp, UPL, and Gateway Pipeline Company, Inc. ("<u>Gateway</u>"). Except as otherwise indicated herein, capitalized terms used but not defined herein have the meanings ascribed to them in the Agreement.

We, as counsel to UtiliCorp and UPL, are rendering this opinion letter to Gateway, its counsel, Bryan Cave LLP, and its lender, [insert name of lender], under Section 7.1(e) of the Agreement.

The documents we have examined in rendering this opinion letter, and upon which we have relied, are the following:

- (a) the Agreement;
- (b) the Release signed by UtiliCorp is favor of UPL on [•], 2001 (the "Release");
- (c) the Transition Services Agreement dated [•], 2001 between UtiliCorp and Gateway (together with the Release and the Agreement the "<u>Transaction Documents</u>");
- (d) UtiliCorp's Certificate of Incorporation and Bylaws, and all amendments thereto;
- (e) UPL's Certificate of Incorporation and Bylaws, and all amendments thereto, and UPL's stock record and minute books;

- (f) MPC's Certificate of Incorporation and Bylaws, and all amendments thereto, and MPC's stock record and minute books; and
- (g) MGC's Certificate of Incorporation and Bylaws, and all amendments thereto, and MGC's stock record and minute books.

In addition, we have examined the originals, or copies certified to our satisfaction, of such other corporate records of UtiliCorp, UPL and its Subsidiaries, certificates of public officials and of the officers of UtiliCorp, UPL and its Subsidiaries, and agreements, instruments, and other documents as we have deemed necessary as a basis for the opinions expressed below. As to questions of fact material to such opinions, we have relied upon certificates of public officials and of the officers of UtiliCorp, UPL and its Subsidiaries and upon representations of UtiliCorp and UPL in the Agreement, without independent investigation or verification.

In making such examination, we have assumed the genuineness of all signatures, the authenticity of all documents submitted to us as originals, and the conformity with the originals of all documents submitted to us as copies. We also have assumed that Gateway has (a) satisfied those legal requirements that are applicable to it under the Transaction Documents to the extent necessary to make the Transaction Documents enforceable against Gateway and (b) complied with all legal requirements pertaining to its status as such status relates to its rights to enforce the Transaction Documents against UtiliCorp and UPL.

Our opinions are limited to the law of the State of Missouri, the General Corporation Law of the State of Delaware and the Federal law of the United States, and we do not express any opinion herein concerning any other law.

Based upon and subject to the other paragraphs hereof, we are of the opinion that:

- 1. Each of UtiliCorp, UPL and each Subsidiary is a corporation duly organized and validly existing in good standing under the laws of the State of Delaware and is duly qualified as a foreign corporation in Missouri, and has the requisite corporate power to carry on its business as presently conducted.
- 2. The execution, delivery and performance by UtiliCorp of the Transaction Documents are within its corporate powers and have been duly authorized by all requisite corporate action on the part of UtiliCorp.
- 3. The Transaction Documents have been duly executed and delivered by UtiliCorp and constitute the legal, valid and binding obligations of UtiliCorp and are enforceable against UtiliCorp in accordance with their terms.
- 4. The execution and delivery of the Transaction Documents, the sale of the Shares and the fulfillment of and compliance with the terms and provisions of the Transaction Documents by UtiliCorp do not (i) violate any of the provisions of the Certificate of Incorporation or By-Laws of UtiliCorp, UPL or any Subsidiary, (ii) to our knowledge, violate any applicable statute, rule or

regulation, or (iii) violate any existing obligation of UtiliCorp, UPL or any Subsidiary under any judgment, ruling, writ, injunction, decree or other court order known to us, which is now applicable to UtiliCorp, UPL or any Subsidiary or any of their respective material properties or assets.

- 5. No authorization, consent or approval of, or filing by UtiliCorp, UPL or any Subsidiary with, any governmental body or authority is necessary for the sale of the Shares or the consummation of the transactions contemplated by the Transaction Documents except as disclosed in Schedule 2.4 of the Agreement.
- 6. UPL's authorized stock consists of 1,000 shares of Common Stock, \$1.00 par value per share, of which 1,000 shares are issued and outstanding. Immediately prior to the consummation of the transactions described in the Agreement, UtiliCorp was the sole registered owner of the Shares and UPL was the sole registered owner of the outstanding stock of each Subsidiary. All of the Shares and all of the stock of each Subsidiary have been duly authorized and validly issued and are fully paid and non-assessable. To our knowledge, there are no contracts, subscriptions, options, warrants, commitments, agreements or rights of any kind to anyone other than Gateway to acquire any stock of UPL or any of its Subsidiaries.

We hereby confirm that there are no actions or proceedings known to us that are pending or overtly threatened in writing against UtiliCorp, UPL or any of its Subsidiaries before any court, governmental agency or arbitrator that seeks to enjoin or otherwise relates to the Transaction Documents.

The opinions expressed herein are subject to the following:

- A. We have assumed for purposes of this opinion that (i) any certifications dated prior to the Closing Date remain true as of the Closing Date; (ii) each certificate of a public official is accurate, complete and authentic and all official public records are accurate and complete; (iii) there has not been any mutual mistake of fact or misunderstanding, fraud, duress or undue influence; (iv) the conduct of the parties to the Transaction Documents has complied with any requirement of good faith, fair dealing and conscionability; (v) there are no agreements or understandings among the parties, written or oral, and there is no usage of trade or course of prior dealing among the parties that would, in either case, define, supplement or qualify the terms of the Transaction Documents.
- B. We have not made any independent investigation or inquiry concerning the business or financial condition of UtiliCorp, UPL or either Subsidiary or concerning the operation, management, use or other dealings with the property of UtiliCorp, UPL or either Subsidiary.
- C. Our opinions as to enforceability are subject to (i) the effect of applicable bankruptcy, insolvency, reorganization, arrangement, moratorium, fraudulent transfer or fraudulent conveyance and other similar laws affecting creditor's

rights generally and judicially developed doctrines relevant to any of the foregoing laws, such as substantive consolidation of entities; (ii) limitations imposed by equitable doctrines, including without limitation, limitations upon the specific enforceability of provisions of the Transaction Documents, concepts of materiality, reasonableness (including commercial reasonableness of the sale or disposition of collateral), good faith and fair dealing, and the availability of injunctive relief or other equitable remedies (regardless of whether considered in a proceeding in equity or at law); and (iii) judicial discretion regarding the determination of damages and entitlement to attorneys' fees and other costs. Enforceability of certain provisions purporting to waive the benefits of statutory provisions or constitutional or common law rights or providing for indemnification, exculpation or release may be limited or otherwise affected by limitations based on statutes, case law or public policy. The opinions in this letter do not include any opinion as to the enforceability of (a) any waiver of jury trial; (b) any choice of law provision; (c) any choice of forum or choice of venue provision; (d) a remedy under certain circumstances where another remedy has been elected; (e) the right of a creditor to use force or cause a breach of the peace in enforcing rights; (f) any power of attorney; (g) the provisions of the Transaction Documents which release, exculpate or exempt a party from, or require indemnification of a party for, liability for its own action or inaction, to the extent the action or inaction involves (I) gross negligence, recklessness, willful misconduct or unlawful conduct or (II) negligence to the extent the provisions are insufficiently explicit; (h) the balance of the Transaction Documents, where part of the Transaction Documents may be unenforceable under circumstances in which the unenforceable portion is an essential part of the agreed upon exchange.

- D. In connection with the opinions expressed in paragraphs 4(ii) and 5 above, we have with your permission relied on the opinions of [insert names of regulatory counsel], copies of which are attached.
- E. When an opinion is based on our knowledge, the opinion or statement means only that we have no actual knowledge to the contrary and does not indicate or imply any investigation or inquiry on our part. For this purpose, "we" means only those attorneys within UtiliCorp's Office of the General Counsel who have done substantive work on this opinion letter.

This opinion is limited to the matters specifically stated in this letter, and no further opinion is to be implied or may be inferred beyond the opinions specifically stated herein. Unless otherwise stated herein, we have made no independent investigation regarding factual matters. This opinion is based solely on the state of the law as of the date of this opinion, and we specifically disclaim any obligation to monitor any of the matters stated in this opinion or to advise the persons entitled to rely on this opinion of any change in law or fact after the date of this opinion which might affect any of the opinions stated herein. This opinion is rendered solely for the benefit of Gateway, its counsel, and its lender in connection with the Agreement, and may not be released to or relied upon by any other person or for any other purpose without our prior written consent.

Very truly yours,

Leslie J. Parrette, Jr.
General Counsel, UtiliCorp United Inc.

#### TRANSITIONAL SERVICES AGREEMENT

THIS TRANSITIONAL SERVICES AGREEMENT is entered into as of [•], 2001, by and between UtiliCorp United Inc., a Delaware corporation ("Seller"), and Gateway Pipeline Company, Inc., a Delaware corporation (the "Company").

WHEREAS, Seller owns all of the issued and outstanding shares of capital stock (the "Shares") of UtiliCorp Pipeline Systems, Inc., a Delaware corporation ("UPL"); and

WHEREAS, the Company desires to purchase and Seller desires to sell the Shares on the terms and conditions set forth in the Stock Purchase Agreement dated February \_\_\_, 2001 (the "Stock Purchase Agreement"); and

WHEREAS, in connection with the sale and purchase of the Shares, Purchaser desires that Seller provide certain transitional services to the Company (the "Services"), and Seller has agreed to provide the Services, in accordance with the terms and conditions of this Agreement;

NOW, THEREFORE, in consideration of the foregoing, the parties hereto agree as follows:

#### ARTICLE I

#### THE SERVICES

- 1.1 The Services. Seller shall provide or cause to be provided to the Company or its subsidiaries, Missouri Pipeline Company ("MPC") and Missouri Gas Company ("MGC"), the Services set forth in Schedule 1.1(a). The Company shall provide or cause to be provided to Seller or its subsidiary, Omega Pipeline Company ("Omega"), the Services set forth in Schedule 1.1(b). The party providing or causing to be provided the Services hereunder shall be referred to herein as the "Service Provider."
- 1.2 Service Parameters. The Service Provider shall provide the Services only to the extent and under the personnel availability conditions that such Services are provided immediately prior to the date this Agreement is executed. Furthermore, the Services will be available only for purposes of supporting the conduct of business substantially in the manner it was conducted immediately prior to the date this Agreement is executed.
- 1.3 Impracticability. The Service Provider shall not be required to provide any Service to the extent the performance of such Service (a) becomes impracticable as a result of a cause or causes outside the reasonable control of the Service Provider, (b) would require the Service Provider to violate any applicable laws, rules or regulations, or (c) would result in the breach of any agreement or other applicable contract.

1.4 Additional Resources. In providing the Services, the Service Provider shall not be obligated to (a) hire any additional employees, (b) maintain the employment of any specific employee, or (c) purchase, lease or license any additional equipment or materials.

#### ARTICLE II

#### TERM AND TERMINATION

- 2.1 Term. The Services shall commence on the date this Agreement is executed and continue for a period of one year.
- 2.2 Termination. The performance of any Service may be terminated by theparty receiving the Service (the "Service Recipient") at any time by providing 20-days prior written notice to the Service Provider. Furthermore, this Agreement or the performance of any Service may be terminated by the mutual written consent of the parties.

#### ARTICLE III

#### COMPENSATION

- 3.1 Charges For Services. The Service Recipient will reimburse the Service Provider for all direct and indirect costs, including out-of-pocket expenses, incurred by the Service Provider in providing the Services, plus an administration fee equal to 2% of such costs.
- 3.2 Payment Terms. The Service Provider shall bill the Service Recipient monthly for all charges pursuant to this Agreement. Such bills shall be accompanied by reasonable documentation supporting such charges. The Service Recipient shall pay the Service Provider for all Services provided hereunder within 15 days after receipt of an invoice therefor. Late payments shall bear interest at the lesser of 18% per annum or the maximum rate allowed by law.

### ARTICLE IV

#### GENERAL OBLIGATIONS; STANDARD OF CARE

- 4.1 Performance Standards. The Service Provider shall use its reasonable commercial efforts to provide the Services in accordance with its policies, procedures, and practices in effect immediately prior to the date this Agreement is executed and to exercise the same degree of care and skill as it exercises in performing similar services for itself.
- 4.2 DISCLAIMER OF WARRANTIES. EXCEPT AS OTHERWISE SET FORTH HEREIN, THE SERVICE PROVIDER MAKES NO WARRANTIES, EXPRESS, IMPLIED OR STATUTORY, INCLUDING BUT NOT LIMITED TO THE IMPLIED WARRANTIES OF MERCHANTABILITY OR FITNESS FOR A PARTICULAR PURPOSE, WITH RESPECT TO THE SERVICES OR OTHER DELIVERABLES PROVIDED BY IT HEREUNDER.
- 4.3 Indemnification by the Service Recipient. The Service Recipient shall indemnify and hold harmless the Service Provider from and against any and all liabilities that arise out of, or result from the provision of the Services by the Service Provider in accordance with this Agreement, other

than liabilities arising from the gross negligence or willful misconduct of the Service Provider or its agents or employees.

4.4 Good Faith Cooperation. The parties will use good faith efforts to cooperate with each other in all matters relating to the provision and receipt of the Services.

#### ARTICLE V

#### RELATIONSHIP BETWEEN THE PARTIES

The relationship between the parties established under this Agreement is that of independent contractors, and neither party shall be deemed an employee, agent, partner, or joint venturer of or with the other. The Service Provider will be solely responsible for the payment of any employment-related taxes, insurance premiums, or employment benefits in respect of the performance of the Services by the Service Provider personnel under this Agreement.

### ARTICLE VI

#### **SUBCONTRACTORS**

The Service Provider may engage one or more subcontractors to perform all or any portion of its duties under this Agreement, provided that the Service Provider remains responsible for the performance of each such subcontractor, and the charges for the Services delegated to a subcontractor shall be the lesser of (a) the amount charged by the subcontractor or (b) the amount that would have been payable to the Service Provider under Article III above if the Service Provider had provided such Services.

#### ARTICLE VII

#### FORCE MAJEURE

The Service Provider will be excused for any failure or delay in performing any of its obligations under this Agreement if such failure or delay is caused by Force Majeure. For the avoidance of doubt, "Force Majeure" means any circumstance or event beyond the reasonable control of the party relying upon such event or circumstance, including, without limitation: any act of God; any accident, explosion, fire, ice, earthquake, lightning, tornado, hurricane, or other severe weather condition or calamity; any civil disturbance, labor dispute, or labor or material shortage; any sabotage or acts of terrorism; any acts of a public enemy, uprising, insurrection, civil unrest, war, or rebellion; or any action or restraint by court order or public or governmental authority or lawfully established civilian authorities.

#### **ARTICLE VIII**

#### **MISCELLANEOUS**

8.1 Entire Agreement. This Agreement and the Schedules attached hereto constitute the entire agreement between the parties with respect to the subject matter hereof and thereof and shall supersede all prior written and oral and all contemporaneous oral agreements and understandings with respect to the subject matter hereof and thereof.

- 8.2 Governing Law. This Agreement shall be governed and construed and enforced in accordance with the laws of the State of Missouri as to all matters, regardless of the laws that might otherwise govern under the principles of conflicts of laws applicable thereto.
- 8.3 Interpretation. The headings contained in this Agreement are for reference purposes only and shall not affect in any way the meaning or interpretation of this Agreement.
- 8.4 Notices. Any notice, demand, offer, request, or other communication required or permitted to be given by either party pursuant to the terms of this Agreement shall be in writing and shall be deemed effectively given the earlier of (a) when received, (b) when delivered personally, (c) one business day after being delivered by facsimile (with receipt of appropriate confirmation), (d) one business day after being deposited with a nationally recognized overnight courier service or (e) four days after being deposited in the U.S. mail, First Class with postage prepaid, and in each case addressed to the attention of the other party's Corporate Secretary at the address of its principal executive office or such other address as a party may request by notifying the other in writing.
- 8.5 Nonassignability; Third-Party Beneficiaries. The Company may not, directly or indirectly, in whole or in part, whether by operation of law or otherwise, assign or transfer this Agreement, without Seller's prior written consent, to any entity other than UPL, MPC or MGC, and any attempted assignment, transfer or delegation without such prior written consent shall be voidable at the sole option of Seller. This Agreement shall be binding upon and inure to the benefit of each party hereto and their respective legal representatives and permitted successors and assigns, and nothing in this Agreement, express or implied, is intended to confer upon any other person any rights or remedies of any nature whatsoever under or by reason of this Agreement.
- 8.6 Severability. If any term or other provision of this Agreement is determined by a nonappealable decision by a court, administrative agency or arbitrator to be invalid, illegal or incapable of being enforced by any rule of law or public policy, all other conditions and provisions of this Agreement shall nevertheless remain in full force and effect so long as the economic or legal substance of the transactions contemplated hereby are not affected in any manner materially adverse to either party. Upon such determination that any term or other provision is invalid, illegal or incapable of being enforced, the parties shall negotiate in good faith to modify this Agreement so as to effect the original intent of the parties as closely as possible in an acceptable manner to the end that the transactions contemplated hereby are fulfilled to the fullest extent possible.
- 8.7 Failure Or Indulgence Not Waiver; Remedies Cumulative. No failure or delay on the part of either party hereto in the exercise of any right hereunder shall impair such right or be construed to be a waiver of, or acquiescence in, any breach of any representation, warranty or agreement herein, nor shall any single or partial exercise of any such right preclude any other or further exercise thereof or of any other right. All rights and remedies existing under this Agreement are cumulative to, and not exclusive of, any rights or remedies otherwise available.
- 8.8 Amendment. No change or amendment will be made to this Agreement except by an instrument in writing signed on behalf of each of the parties hereto.
- 8.9 Counterparts. This Agreement may be executed in one or more counterparts, all of which shall be considered one and the same agreement, and shall become effective when one or more counterparts have been signed by each of the parties and delivered to the other parties.

IN WITNESS WHEREOF, the parties have signed this Transitional Services Agreement effective as of the date first set forth above.

UTILICORP UNITED INC.
By:Title:
GATEWAY PIPELINE COMPANY, INC.
By:Title:

### SCHEDULE 1.1(a)

# SERVICES PROVIDER BY SELLER

Seller shall provide the following Services to the Company:

- (a) 24-hour remote surveillance control (SCADA) services, as provided before the closing of the transactions contemplated by the Stock Purchase Agreement;
- (b) The daily administration of nominations, as provided before the closing of the transactions contemplated by the Stock Purchase Agreement;
- (c) Engineering services, as provided before the closing of the transactions contemplated by the Stock Purchase Agreement; and
- (d) Any other service mutually agreed to by Seller and the Company.

## SCHEDULE 1.1(b)

### SERVICES PROVIDED BY THE COMPANY

The Company shall provide to Seller (a) the services provided by UPL, MPC, or MGC to or for the benefit of Omega prior to the closing of the transactions contemplated by the Stock Purchase Agreement and (b) any other services mutually agreed to by Seller and the Company.

5 ر

### SCHEDULE 1.1(a)

### SERVICES PROVIDER BY SELLER

Seller shall provide the following Services to the Company:

- (a) 24-hour remote surveillance control (SCADA) services, as provided before the closing of the transactions contemplated by the Stock Purchase Agreement;
- (b) The daily administration of nominations, as provided before the closing of the transactions contemplated by the Stock Purchase Agreement;
- (c) Engineering services, as provided before the closing of the transactions contemplated by the Stock Purchase Agreement; and
- (d) Any other service mutually agreed to by Seller and the Company.

## SCHEDULE 1.1(b)

## SERVICES PROVIDED BY THE COMPANY

The Company shall provide to Seller (a) the services provided by UPL, MPC, or MGC to or for the benefit of Omega prior to the closing of the transactions contemplated by the Stock Purchase Agreement and (b) any other services mutually agreed to by Seller and the Company.

• . 

### SCHEDULE 1.5(a)

# WORKING CAPITAL ACCOUNTS

	EXAMPLE 12/31/2000
Current Assets	TOTAL
Cash and Cash Equivalents	
Working Funds	\$150.00
Accounts Receivable, Net	4750.00
Customer Accounts Receivable	576,402.66
Other Accounts Receivable	0.00
Prepayments and Other	
Notes Receivable	
City of Cuba, MO (expires May 2001)	6,559.29
Inventory and Supplies	0.00
Total Current Assets	\$583,111.95
Current Liabilities	
Accounts Payable	\$50.00
Accrued Liabilities	
Other Accrued Liabilities	
Accrued Payroll*	8,176,00
Accrued Taxes	•
Accrued Property Taxes**	88,451.28
Total Current Liabilities	96,677.28
NET WORKING CAPITAL	\$486,434.67

<sup>\*</sup> Accrued Payroll represents current payroll and payroll taxes payable at closing for hours worked prior to closing. Seller is responsible for all payroll benefits accrued prior to closing and will retain all operating reserves associated with the payment of such benefits.

<sup>\*\*</sup>Accrued Property Taxes will be the pro-rata portion of property taxes payable in December 2001 for the number of days Seller holds Company prior to Closing compared to 365 days. The Estimated Purchase Price Certificate will reflect an accrued property tax value equal to accruals consistent with property tax payments made in calendar year 2000.

8

•

#### SELLER'S DISCLOSURE SCHEDULE

Stock Purchase Agreement
UtiliCorp United Inc.,
UtiliCorp Pipeline Systems, Inc.,
and
Gateway Pipeline Company, Inc.

February  $\perp$ , 2001

This is the "Seller's Disclosure Schedule" referred to in the Stock Purchase Agreement, dated as of February 1, 2001 (the "Agreement") between UtiliCorp United Inc., a Delaware corporation (the "Seller"), UtiliCorp Pipeline Systems, Inc. (the "Company"), a Delaware corporation, and Gateway Pipeline Company, Inc., a Delaware corporation (the "Purchaser"). This Seller's Disclosure Schedule is part and parcel of the Agreement.

Capitalized terms not defined herein have the meanings ascribed to them in the Agreement. The disclosures in this Seller's Disclosure Schedule are made in connection with the representations and warranties of the Seller and certain covenants of the Seller contained in the Agreement without taking into consideration any standard of materiality applicable to such representations, warranties or covenants. Accordingly, no disclosure made herein shall (i) constitute an admission or determination that any fact or matter so disclosed is material to the business, properties, assets, liabilities, condition (financial or otherwise) or results of operations of the Seller or (ii) be deemed to modify in any respect the standard of materiality applicable to any representation, warranty, covenant or other provision contained in the Agreement.

## SCHEDULE 2.1(a)

- 1. Missouri Pipeline Company, a Delaware corporation.
- 2. Missouri Gas Company, a Delaware corporation.

- 1. The approval of, or the waiver of any approval right by, the Iowa Utilities Board.
- 2. The approval of, or the waiver of any approval right by, the Missouri Public Service Commission.

Portions of the pipeline owned and operated by the Company or its Subsidiaries may not be in compliance with Laws relating to pipeline depth.

None

Condemnation proceeding of Missouri Pipeline Company versus Arla E. Reed and Susan G. Reed in Franklin County Circuit Court; Case No. CV 191-1336CC. Seller agrees that if this case is not settled or otherwise resolved prior to the Closing Date, the Purchase Price shall be reduced by \$15,000.

The Company is a named defendant in <u>Grynberg v UtiliCorp</u>, et al, also know as the "Natural Gas Qui Tam Litigation." This case is part of a large Multi-District Litigation proceeding in Federal Court in Wyoming. Plaintiff Jack Grynberg has brought suit on behalf of the United States under the Federal False Claims Act against several hundred companies in the natural gas industry. Plaintiff alleges that the defendants regularly and repeatedly undermeasure the volume and heating content of natural gas produced on federal and Indian lands, thereby underpaying the royalties due the federal government. Although the case has been pending for several years, it is still in its early procedural stages. All defendants filed motions to dismiss the case. Those motions were argued on March 17, 2000, and have not been ruled on by the court. No discovery has taken place.

The Company is also a named defendant in <u>Quinque v UtiliCorp</u>, et al. In this case plaintiffs have alleged a nationwide class action on behalf of all royalty owners, overriding royalty owners, and working interest owners of every non-federal gas well in the country since 1974. Defendants include approximately 235 companies in the natural gas industry. As in the <u>Grynberg</u> cases, plaintiffs allege the defendants have undermeasured all natural gas produced from these wells. The case was originally filed in state court in Stevens County, Kansas. The case was immediately removed to Federal Court, then transferred to the same Wyoming Multi-District Litigation as the <u>Grynberg</u> cases. In the meantime, however, plaintiffs have filed a motion to remand the case to state court in Kansas. The Wyoming Federal Judge has granted that motion and it now appears the case will be remanded back to state court in Kansas.

#### SCHEDULE 2.12(a)

### **Interconnect Agreements**

See attached list.

#### **Operational Balance Agreements**

See attached list.

#### **Transportation Contracts**

See attached list.

# Franchises, Licenses, Royalty or Other Agreements Respecting or Restricting Intellectual Property Rights.

- 1. Licenses for Scada software packages Intellusion, Genesis, and Bristol Babcock Accol are held by the Company or its Subsidiaries and such licenses will continue to be held by the Company or its Subsidiaries following the Closing.
- 2. Licenses for existing office software packages (i.e., Excel, Word, Powerpoint) are held by Seller and are not being transferred or assigned to Purchaser in this transaction.
- 3. The Company and the Subsidiaries use AutoCad software, the licenses to which are held by Seller and are not being transferred or assigned to Purchaser in this transaction.

# Agreements Made Other Than in the Ordinary Course of the Company's or Any Subsidiary's Business

- 1. Reimbursement agreement dated June 1, 1995 with the City of Cuba, Missouri for repayment of the costs incurred by Missouri Gas Company for the provision and installation of city gate facilities, with final payment due on May 15, 2001. See attached agreement.
- 2. Postage Meter St. Peters Office
- 3. Copier St. Peters Office
- 4. Drinking Water Service St. Peters Office
- 5. Nitrogen and Test Gas Bottles
- 6. See attached list of discount letters and agreements.

Agreements with Any Affiliate of the Company, Except Agreements Among Only the Company and Any of its Subsidiaries

See the attached list of transportation contracts referenced above which includes agreements with affiliates.

## Interconnect Agreements

Missouri Pipeline - Panhandle Eastern

Interconnect Letter signed:

November 28, 1989

Panhandle Eastern: John A. Alholm, Vice President
Missouri Pipeline Company: Tom M. Taylor, President

Missouri Pipeline - Fidelity Natural Gas

June 16, 1992

MPC: Tom M. Taylor

Fidelity: John Davis

Missouri Pipeline - Laclede Gas

August 14, 1996

MPC: Richard C. Kreul Laclede Gas: Kenneth J. Neises

Missouri Pipeline - Union Electric / Winfield

October 17, 1992

MPC: Richard C. Kreul U. E.: William J. Carr

Missouri Pipeline - Union Electric / Route N

July 14, 1999

MPC: Richard C. Kreul

Ameren U.E.: Senior Vice President

Missouri Gas Co. - City of Waynesville

February 10, 1994

MGC: Richard C. Kreul

City of Waynesville: Bill Randall

Missouri Gas Co. - City of St. Robert

July 30, 1993

MGC: Richard C. Kreul

City of St. Robert: Donald Scott

Missouri Gas Co. - City of Cuba

April 23, 1993

MGC: Richard C. Kreul

City of Cuba: Ray Mortimeyer

Missouri Gas Co. - City of St. James

July 10, 1992

MGC: Tom M. Taylor

City of St. James: Nelson Hart

Missouri Gas Co. -Omega for

Fort Leonard Wood

July 29, 1993

MGC: Richard C. Kreul

Omega Pipeline: Phil L. Miller

Missouri Gas Co - City of Rolla

August 22,1994

MPC: Richard C. Kreul

MoPub: Michael Apprill

Missouri Gas Co. - City of Rolla and Salem

September 1, 1996

MPC: Richard C. Kreul

MoPub: Harry F. Ono

## Operational Balance Agreements

Missouri Pipeline - Fidelity/Sullivan

July 23, 1992

MPC: Tom M. Taylor

Fidelity: Kenneth Matzdoff

Missouri Pipeline - Laclede Natural Gas

November 8, 1989

MPC: Tom M. Taylor

Laclede Gas: Robert. C. Jaudes

Missouri Pipeline - Panhandle Eastern

June 27, 1997

MPC: Dolin Argo, Director Busness Operations

PEPL: John Coppers, Director, Marketing Operations

Prepared: June 5, 2000

File name: Word/Interconnect Agreement and OBA Agreement Lists

# MISSOURI PIPELINE COMPANY TRANSPORTATION CONTRACTS

## MISSOURI GAS COMPANY TRANSPORTATION CONTRACTS

# NON-PROPRIETARY

## **UPL DISCOUNTS**

# Non-proprietary

## REIMBURSEMENT AGREEMENT

Exhibit A

Interconnect Facility Operating Agreement-

kc1-3652.1

5

AN ORDINANCE AUTHORIZING THE MAYOR OF THE CITY OF CUBA, MISSOURI, ON BEHALF OF THE CITY OF CUBA, MISSOURI, TO ENTER INTO AN AGREEMENT WITH MISSOURI GAS COMPANY, REFINANCING MONIES OWED BY THE CITY TO MISSOURI GAS COMPANY PURSUANT TO THE INTERCONNECT FACILITY OPERATING AGREEMENT AND PROVIDING FOR AN EFFECTIVE DATE OF THIS ORDINANCE.

WHEREAS, the Board of Aldermen of the City of Cuba, Missouri, has determined that it is in the best interest of the City of Cuba, Missouri, to implement a plan for payment of monies owed to Missouri Gas Company pursuant to the Interconnect Facility Operating Agreement, executed by the City of Cuba, Missouri, for the natural gas system currently operated by the City; and

WHEREAS, Missouri Gas Company has proposed certain terms set out in the Agreement attached hereto as Exhibit "A"; and

WHEREAS, the terms of payment set out in said Agreement are in the best interest of the City of Cuba, Missouri.

NOW, THEREFORE, BE IT ORDAINED BY THE BOARD OF ALDERMEN OF THE CITY OF CUBA, MISSOURI, AS FOLLOWS:

Section 1: The City of Cuba, Missouri, shall enter into an Agreement with Missouri Gas Company, a copy of said Agreement being attached hereto as Exhibit "A".

Section 2: The Mayor of the City of Cuba, Missouri, is authorized to execute the Agreement on behalf of the City of Cuba, Missouri.

Section 3: This Ordinance shall be effective from and after its date of passage and approval.

READ TWO TIMES AND PASSED BY THE BOARD OF ALDERMEN OF THE CITY OF CUBA, MISSOURI, THIS 3rd DAY OF July 1995.

ROBERT H. STEWART, MAYOR

ATTEST:

MELINDA JAMES, CITY CLERK

(CITY SEAL)

# Mon-proprietary

SCHEDULE 2.12(d)

## SCHEDULE 2.12(e)

None

### **SCHEDULE 2.13A**

- 1. 1997 Ford F-150 4X4 Super Cab Vehicle No. 709701. VIN: 1FTDX1867VKA76323
- 2. 1999 Ford F-150 4X4 Super Cab Vehicle No. 839145. VIN: 1FTRX18WOXDB84601
- 3. 1999 Ford SD F250 4X4 Reg Cab Vehicle No. 848011. VIN: 1FTNF21FOXE811406
- 4. 1999 Ford F-150 4X4 Super Cab Vehicle No. 839143. VIN: 1FTRX18W6XDB84599
- 5. 1997 Ford F-150 4X4 Super Cab Vehicle No. 709702. VIN: 1FTDX1869VKA76324
- 6. 1996 Ford F-250 HD 4X4 Reg Cab Vehicle No. 709601. VIN: 2FTHF26F6TCA53586
- 7. 1994 Ford F-350 4X4 Chassis Cab with Utility Bed Hydraulic Crane Vehicle No. 709401. VIN: 1FDKF38F8RNB17371
- 8. 1993 16' Flat Bed Trailer 4 X Utility VIN: JMC7U7329P1SM2310
- 9. 1996 16' Flat Bed Trailer 4 X Utility. VIN: 1S9UJ1620T1513168

#### **SCHEDULE 2.13B**

### St. Peter's Property

- 1. Building lines and easements according to plat of said subdivision and building lines, easements and restrictions according to instrument recorded in Book 807 page 226.
- 2. Deed of Trust executed by Esco Energy, Inc., Aleco Production Company, Esco Exploration, Inc., Missouri Pipeline Company, Omega Pipeline Company and Vesta Energy Company and United States Trust Company of New York, and Patrick J. Kiely, as Trustees for First National Bank and Trust Company of Tulsa, dated June 15, 1990, recorded August 9, 1990, in Book 1334 page 1, for \$15,000,000.00.
- 3. Deed of Trust executed by Esco Energy, Inc., Aleco Production Company, Esco Exploration, Inc., Missouri Pipeline Company, Omega Pipeline Company and Vesta Energy Company and United States Trust Company of New York, and Patrick J. Kiely, as Trustees for First National Bank and Trust Company of Tulsa, dated June 15, 1990, recorded June 21, 1990, in Book 1326 page 570, and supplements thereof recorded in Book 1342 page 1270, Book 1406 page 761 and Book 1472 page 1820, for \$15,000,000.00.

Note: The Deeds of Trust referenced as items 2 and 3 shall be removed prior to closing.

### Washington Property

- 1. Building lines and easements as per the plat of said property.
- 2. Surveyor's Record 1 Page 16.
- 3. Easements to Union Electric Company, for utility purposes, according to deeds recorded in Book 85 Page 455, Book 128 Page 248 and Book 138 Page 582.
- 4. Easement Contract (Option) recorded in Book 636 Page 606 recorded May 31, 1991 and Book 630 Page 974 recorded April 16, 1991.
- 5. Easement to Laclede Gas Company as recorded in Book 700 Page 44, and Book 1285 Page 1198

Note: Record title to the St. Peters and Washington properties is currently in the name of Missouri Pipeline Acquisition Corp., and will be corrected to Missouri Pipeline Company prior to closing.

#### All Properties:

Prior recorded deeds of trust executed by property owners along Company and Subsidiary right-of-ways.

### SCHEDULE 2.13(a)(i)

#### Office Leases

UtiliCorp Pipeline Systems and MJSM Partnership St. Peters office Office Lease dated August 25, 2000. One-year term. Monthly rent \$1,656.75.

UtiliCorp Pipeline Systems and Missouri Public Service Missouri Gas Company – Rolla office Lease arrangement dividing warehouse space Monthly rent \$375.00 – includes electric, gas, water, trash. Month-to-month

(See leases attached)

2000

2001

Dollars

#### COMMERCIAL LEASE

This Lease, made and entered into, this 25 day of August 2000 by and between MJSM Partnership

Parties

hereinafter called Lessor, and UtiliCorp Pipeline Company

hereinafter called Lessee,

WITNESSETH, That the said Lessor for and in consideration of the rents, covenants and agreements bereinnster mentioned and hereby agreed to be paid, kept and performed by said Lessee, or Lessees, successors and assigns, has leased and by those presents does lesse to said Lessee the following described premises, situated in the City of St. Peters State of Missouri, to-wit:

Premises

Cherokee Industrial Park 110-Algana Court St. Peters, MO 63376

3,000 sq. ft. Plus 800 sq. ft. Upstairs office space.

Use of Premises

To have and to hold the same, subject to the conditions herein contained, and for no other purpose or business than that of

Office/warehouse for pipeline equipment and/or related items.

Term and Rental for and during the term of 2 years

commencing on the 1<sup>st</sup> August ...

and ending on the 31<sup>st</sup> July ...

at the yearly rental of Nineteen Thousand Eight Hundred Eighty One

payable in advance in equal monthly installments of \$1656.75

Payment to be mailed to: MJSM Partnership c/o Susan Weissinger 63 Klaas Lane Foristel, MO 63348

Lessee has the option to renew the lease at the end of the term for the base rent plus a cost of living increase. Consumer Price Index - All Urban Consumers.

on the 1st day of each and every mouth during the said term.

Assignment assigned or of the Subletting bankr credit

This Lease is not assignable, nor shall said premises or any part thereof be sublet, used or permitted to be used for any other purpose other than above set forth without the written consent of the Lesser endorsed hereon; and if this Lease is assigned or the premises or any part thereof sublet without the written consent of the Leaser, or if the Lessee shall become the subject of a court proceeding in bankruptcy or liquidating receivership or shall make an assignment for the benefit of creditors, this Lease may by such fact or unauthorized act be canceled at the option of the Lessor. Any Assignment of this Lease or subletting of said premises or any part thereof with the written consent of the Lessor shall not operate to release the Lessee from the fulfillment of Lessee's part of the covenants and agreements herein contained to be by said Lessee performed, nor authorize any subsequent assignment or subletting without the written consent of the Lessor.

Répairs and Alterations

> All repairs and alterations deemed necessary by Lessee shall be made by said Lessee at Lessee's cost and expense with the consent of Lessor; and all repairs and alterations so made shall remain as a part of the realty; all plate and other glass now

יוֹטָריק מעשע/מעער באבו־וני־סונס די מאַני־וני־סונס די מעשע/מעער

in said demised premises is at the risk of said Lessee, and if broken, is to be replaced by and at the expense of said Lessee.

The Lessor reserves the right to prescribe the form, size, character and location of any and all awnings affixed to and all signs which may be placed or painted upon any part of the demised premises, and the Lessee agrees not to place any awning or sign on any part of the demised premises without the written consent of the Lessor, or to bore or cut into any column, beam or any part of the demised premises without the written consent of the Lessor. The Lessee and all holding under said Lessee agrees to use reasonable diligence in the care and protection of the premises during the term of this Lease, to keep the water pipes, sewer drains, heating apparatus, elevator machinery and sprinkler system in good order and repair and to surrender said promises at the termination of this Lease in substantially the same and in as good condition as received, ordinary wear and tear excepted.

The Lessee shall pay according to the rules and regulations of the water department for all water used on the domised premises. The Lessee will erect fire escapes on said premises at said Lessee's own cost, according to law, should the proper authorities demand same.

The Lessee agrees to keep said premises in good order and repair and free from any nuisance or filth upon or adjacent thereto, and not to use or permit the use of the same or any part thereof for any purpose forbidden by law or ordinance now in force or hereafter enacted in respect to the use or occupancy of said premises. The Lessor or legal representatives may, at all reasonable hours, enter upon said premises for the purposes of examining the condition thereof and making such repairs as Lessor may see fit to make.

Lessor shall procure and maintain at its cost an insurance policy insuring the store building and it's improvements against loss or damage by fire, lighting and all standard extended coverage, perils and vandalism for a minimum of 80% co-insurance based on actual cash value thereof. Lessee at its solo cost and expense shall keep all its trade fixtures, equipment, its improvements and inventory on the Leased premises insured against loss or damage of fire, extended coverage, perils and vandalism for the full fair insurable value thereof.

The Lessee, at its cost, shall obtain and continue in effect General Liability insurance, naming the Lessor as an "additional insured, for its interest as Lessor", protecting against liability arising out of use and operations of the Leased premises. Said insurance, for Bodily Injury and Property Damage shall have a combined single limit of liability of no less than one million dollars, (\$1,000,000,000) per each occurrence.

Lessor hereby releases Lessee, any assignee and sublessee of Lessee, and Lessee hereby agrees to release Lessor or to secure its sublessee's release of Lessor, from and against any and all claims, demands, liabilities or obligations whatsoever for damage to the property of either Lessor, Lessee or its sublessee resulting from or in anyway connected with any fire, accident or other casualty, whether or not such fire, accident or other casualty shall have been caused by the negligence or contributory negligence of Lessor or Lessee, any assignee or sublessee of Lessor and Lessee or by agent, associate or employee of either of them, to the extent that such damage or loss either is insured against under any insurance contract which at the time of such damage or loss permits waiver of subrogation rights prior to a loss thereunder, or is to be insured against by the provisions of the prior paragraph above.

If the cost of insurance to said Lessor on said premises shall be increased by reason of the occupancy and use of said demised premises by said Lessee or other person under said Lessee, all such increase over the existing rate shall be paid by said Lessee to said Lessor on demand. The Lessee agrees to pay double rent for each day the Lessee, or anyone holding under the Lessee, shall rotain the demised premises after the termination of this Lesse, whether by limitation or forfeiture.

Lessor shall not be liable to said Lessee or any other person or corporation, including employees, for any damage to their person or property caused by water, rain, snow, frost, fire, storm and accidents, or by broakage, stoppage or leakage of water, gas, heating and sewer pipes or plumbing, upon, about or adjacent to said premises.

The destruction of said building or premises by fire, or the elements, or such material injury thereto as to render said premises unquestionably untenantable for 30 days, shall at the option of said Lessor or Lessee produce and work a termination of this Lesse.

If the Lessor and Lessee cannot agree as to whether said building or premises are unquestionably untenantable for 30 days, the fact shall be determined by arbitration; the Lessor and the Lessee shall each choose an arbitrator within five days after either has notified the other in writing of such damage, the two so chosen, before entering on the discharge of their duties, shall elect a third, and the decision of any two of such arbitrators shall be conclusive and binding upon both parties hereto.

If it is determined by arbitration, or agreement between the Lessor and the Lessee that said building is not unquestionably untenantable for days, than said

Damage to Tenants Property Lessor shall restore said building at Lessor's own expense, with all reasonable speed and promptness, and in such case a just and proportionate part of said rental shall be abated until said premises have been restored.

Failure on the part of the Lessee to pay any installment of rept or increase in insurance rate promptly as above set out, as and when the same becomes due and payable, or failure of the Lessee promptly and faithfully to keep and perform each and every covenant, agreement and exipulation herein on the part of the Lessee to be kept and performed, shall at the option of the Lessor cause the forfeiture of this Lesse.

Possession of the within demised premises and all additions and permanent improvements thereof shall be delivered to Lessor upon ten days' written notice that Lessor has exercised said option, and thereupon Lessor shall be entitled to and may take immediate possession of the demised premises, any other notice or demand being hereby waived.

Any and all notices to be served by the Lessor upon the Lessoe for any breach of covenant of this Lesse, or otherwise, shall be served upon the Lessee in person, or left with anyone in charge of the premises, or posted upon some conspicuous part of said premises.

Re-Entry

Said Lessee will quit and deliver up the possession of said premises to the Lessor or Lessor's being, successors, agents or sessigns, when this Lesse terminates by limitation or forfeiture, with all window glass replaced, if broken, and with all keys, locks, bolts, plumbing fixtures, elevator, sprinkler, boiler and heating appliances in as good order and condition as the same are now, or may hereafter be made by repair in compliance with all the covenants of this Lesse, save only the wear thereof from reasonable and careful use.

But it is hereby understood, and Lessee hereby covenants with the Lessor, that such forfeiture, annulment or voldance shall not relieve the Lesses from the obligation of the Leases to make the mouthly payments of rent hereinbefore reserved, at the times and in the manner aforesaid; and in case of any such default by the Lesses, the Lesser may re-let the said premises as the agent for ead in the name of the Lesses, at any rental readily obtainable, applying the proceeds and the avails thereof, first, to the payment of such expanse as the Lessor may be put to in re-entering, and then to the payment of said rent as the some may from time to time become due, and toward the fulfillment of the other covenants and agreements of the Lessee berein contained, and the balance, if any, shall be paid to the Lessee; and the Lessee hereby covenants and agrees that if the Lessor shall recover or take possession of said premises as aforesaid, and be unable to re-let and rent the same so as to realize a sum equal to the rent hereby reserved, the Lessee shall and will pay to the Lessor any and all loss of difference of rent for the residue of the term. The Lessee hereby gives to the Lesser the right to place and maintain its usual "for rent" signs upon the demised premises, in the place that the same are usually displayed on property similar to that herein demised, for the last thirty days of this lease.

No Constructive Waiver

No waiver of any forfeiture, by acceptance of rent or otherwise, shall waive any subsequent cause of forfeiture, or breach of any condition of this lease; nor shall any consent by the Lessor to any assignment or subletting of said premises, or any part thereof, be held to waive or release any assignee or sub-lease from any part of the foregoing conditions or covenants as against him or them; but every such assignee or sub-leases shall be expressly subject thereto.

Whonever the word "Lessor" is used herein it shall be construed to include the heirs, executors, administrators, successors, sesigns or legal representatives of the Lessor; and the word "Lessoe" shall include the heirs, executors, administrators, successors, assigns or legal representatives of the Lessoe and the words Lessor and Lessoe shall include single and planal, individual or corporation, subject always to the restrictions herein contained, as to subjecting or assignment of this Lessoe.

IN WITNESS WHEREOF, the said parties aforesaid have duly executed the foregoing instrument or caused the same to be executed the day and year first above written.

anager MO.

Leuse LESSOR

My term expires:

Notary Public

31-131-1363

بالأن و أ

ולים בישולנים ביישו

10750 East 350 Highway, Suite 310 P.O. Box 37138 Kansas City, Missouri 64138 818-737-7900 Fex: 816-737-7925

## UTILICORP PIPELINE SYSTEMS

**ENERGY ONE** 

June 29, 2000

Missouri Public Service Mr. Dennis Bennett 1006 N. Pine Rolla, MO 65401

Dear Dennis:

This letter is intended to detail the agreement between Missouri Gas Company ("MGC") and Missouri Public Service ("MPS") with regard to the rental of space by MGC from MPS.

MGC is renting space from MPS at 11824 Private Drive, #2005, Rolla, MO 65401. The monthly rental amount is \$375.00, which includes all utilities, except telephone. MGC pays MPS this amount on a monthly basis.

If the above description meets with your understanding of our agreement, please sign below. Retain one copy for your records and return a copy to me.

Dennis, thank you for working with the pipeline in providing facilities in the MPS Rolla warehouse.

Sincerely,

Mike Megaris

Director, Business Operations Natural Gas

CC: Rick Kreul

Dave Wallen

mile Myse